Dynamic Risk Assessment, Personality Disorder, and Key Developmental Variables in Sexual Offenders

by

Rebecca-Louise Coles

A thesis submitted to the College of Life and Environmental Sciences of the University of Birmingham

for the degree of

Doctorate in Forensic Psychology Practice (ForenPsyD)

Centre for Forensic and Criminological Psychology
School of Psychology
The University of Birmingham

February 2011



University of Birmingham Research Archive e-theses repository

This unpublished thesis/dissertation is copyright of the author and/or third parties. The intellectual property rights of the author or third parties in respect of this work are as defined by The Copyright Designs and Patents Act 1988 or as modified by any successor legislation.

Any use made of information contained in this thesis/dissertation must be in accordance with that legislation and must be properly acknowledged. Further distribution or reproduction in any format is prohibited without the permission of the copyright holder.

THESIS ABSTRACT

This thesis examined the relationship between key developmental variables, dynamic risk factors, and personality. First, a detailed review of the literature pertaining to these areas is presented, a narrative review with systematic search strategies was compiled due to the breath of the topic areas. Second, an empirical research project was conducted to investigate the presence of any relationships between key developmental variables, dynamic risk factors on the Stable-2007 (formally Stable-2000 and Sex Offender Needs Assessment Rating) (Hanson & Harris, 2001; Hanson, Harris, Scott, & Helmus, 2007), and personality disorders measured using the Million Multiaxial Clinical Inventory (MCMI-III) (Millon, Millon, Davis, & Grossman, 1997) and the any relationship to treatment attrition. The Challenge project data was utilised for this research which comprised information on 106 sex offenders both child molesters (n=69) and rapists (n=37). The research demonstrated number of relationships between personality disorder, key developmental variables, and the Stable-2007 items. There were very few significant associations between any of the variables and treatment attrition. Third, a case study aimed to demonstrate the practical utility of Stable 2007 discussed in Chapter 1 and researched in Chapter 2 is presented and relevant interventions are discussed. Fourth, a critique of a psychometric measure, the Stable-2007 is provided, which demonstrates continuing advances and validation of dynamic risk assessments. Finally, an overall discussion of each of the chapters is provided.

ACKNOWLEDGEMENTS

Many people from the University of Birmingham have been an asset throughout my time on the Forensic Psychology Doctorate course, including supervisors and professors. In particular, I would like to thank Professor Anthony Beech for his guidance. His wealth of experience and knowledge in this area has been unspeakably valuable to the completion of this thesis. Special thanks to Sue Hanson for her ongoing support and incredible knowledge. I would also like to thank the centre for their input both into this thesis and into my professional development. Specifically, I would like to thank Dr. Jackie Craissati for her knowledge, perseverance, support, and patience. Personally, I would like to thank my family and friends, as without them I would never have had the strength and determination to complete this thesis. Finally, I would like to dedicate this thesis to the late Frederick Coles, my grandfather, my inspiration, my role model.

Contents

Chapter Number	Page Number
Introduction:	1
Chapter 1: The links between dynamic risk factors, personality disorders, and key variables: A Narrative Review.	6 y developmental
Chapter 2: Investigating the relationship between key developmental variab disorder, and dynamic risk assessment in sexual offenders: An Empiric	•
Chapter 3: A Single Case Study: A 41-year-old high risk sexual offender.	69
Chapter 4: Critique of a Psychometric Measure: Stable-2000	126
Chapter 5: Discussion	141
References:	158
Appendices:	164

List of Tables

Chapter Number	Page Number
Chapter 1:	
None	
Chapter 2:	
Table 1: Background and offence related variables for the samp	ple 51
Table 2: Results of Psychometric measures for risk	53
Table 3: Results of the MCMI-III	55
Table 4: ROC results	61
Chapter 3:	
Table 1: Functional analysis of Nigel	107
Chapter 4:	
None	

List of Appendices

Chapter Number	Page Number
Chapter 1	
Appendix 1: List of Cochrane References	164
Appendix 2: Search Strategies	171
Chapter 2	
Appendix 3: Key Developmental Variables	173
Appendix 4: Measures	174
Chapter 3	
None	
Chapter 4	
Appendix 5: Stable-2007 Tally Sheet	176
Chapter 5	
Appendix 6: Personality Disorder Clusters	178

INTRODUCTION

The overall aim of this thesis was to investigate the relationships between dynamic risk, factors, personality disorder, and key developmental variables in sexual offenders. This aimed to increase the understanding on these variables and to discuss how they may be measuring the same underlying constructs. It aims to contribute to the overall understanding of sexual offender risk assessment and discuss the need for more holistic risk assessment processes considering a range of background and dynamic variables.

Background

The field of sex offender risk assessment has been a topic of much study and debate in recent decades. This has wide implications on society and victims and as such researchers and clinicians are continually aware of the importance of accurate assessments. The development of risk assessment tools have gone through a number of generations from unstructured clinical judgement through actuarial tool such as the Static-99 (Hanson & Thornton, 2000) and dynamic measures such as the Stable 2007 (Hanson, Harris, Scott, & Helmus, 2007). Actuarial assessments continue to show at least moderate predictive accuracies, AUC ROC >0.7 (Hanson, 2004). However, critics assert that actuarial tools, among other flaws, do not allow for the assessment of offender need and cannot measure change. Dynamic risk assessments have began to overcome this criticism and there is ongoing research into the combination of static, stable dynamic, and acute dynamic risk tools (Hanson, Harris, Scott, & Helmus, 2007). However, further validation of such measures needs to be conducted.

An important component of developing effective sex offender risk assessment tools is well researched risk variables. There are evidently common themes in the risk assessments such as age and previous offending. However, other risk factors remain ambiguous such as that of personality and psychopathy (Hanson, 2000; Hanson & Morton-Bourgon, 2004). Personality disorder, largely antisocial personality disorder and psychopathy have been found to be predictive of sexual offender recidivism and there is a high prevalence of these variables in the sex offender population (Hanson and Bussierre, 2003; Langstrom, Sjostedt, and Grann, 2004; Hanson & Harris, 1998, 2000; Hildebrand, de Ruiter & de Vogel, 2004; Quinsey, Rice, & Harris, 1995; Rice, Harris, & Quinsey, 1990. However, it is evident that ongoing research needs to be conducted in order to clarify the role of personality disorder in sexual offence recidivism. In addition, there is been more recent research into the role of developmental or background variables in sex offender risk and recidivism (e.g., Craissati & Beech, 2006). This has been related to poor childhood attachments (e.g., Ward, Hudson, & Marshall, 1996) and later personality development (Marshall, Hudson, & Hodkinson, 1993). Furthermore, Craissati and Beech (2006) provide some evidence for a model of risk assessment using developmental variables. Finally, these risk related factors have been related to sexual offender treatment attrition (e.g., Craissati, Webb, & Keen, 2005). The research in this area is relatively new and sparse. However, it is a notable topic.

Chapter 1

Chapter one presents a narrative literature review with a systematic search procedure. This aims to present the research on the relationships between dynamic risk factors, personality disorder, and key developmental variables and the links between these factors and treatment adherence in sexual offenders.

Chapter 2

Chapter two presents an empirical research project using data from the Challenge Project (a community assessment and treatment programme for sexual offenders). This aims to build on previous research to determine the links between dynamic risk factors using the Stable-2007, personality disorder using the MCMI-III, and key developmental variables. The research also seeks to investigate the relationship of these variables to treatment attrition.

Chapter 3

Chapter three presents a single case study of a 41-year-old psychopathic rapist being assessed for treatment on the Challenge Project. A full assessment including clinical interview, psychometric assessment, risk assessment, and formulation is presented. Due to the setting a treatment could not be offered by the author. However, recommendations for treatment on the Challenge Project are presented. The case study highlights the importance of individual assessment and formulations within the context of the relevant theory and empirical research. It also demonstrates the importance of considering personality as a risk factors and a factor for treatment considerations.

Chapter 4

Chapter four presents a critique of a psychometric measure relevant to the empirical research, namely the Stable-2000 (Hanson & Harris, 2001). This is a stable dynamic risk assessment. The critique provides a description of the tool and discusses the reliability and validity, as well as the need for further validation for use in forensic settings.

Chapter 5

Chapter five concludes the thesis, discussing the overall main findings. The implications and limitations of the thesis are discussed as well as its applicability to further research and clinical practice.

ETHICAL CONSIDERATIONS

The ethical guidelines outlined by the British Psychological Society were considered and adhered to throughout this thesis. In Chapter 1, the Challenge research project had existing ethical approval. This study was also reviewed at the practices Board of Directors meeting and was approved through the University of Birmingham. In Chapter 3, Nigel provided verbal consent for the information to be used in this manner and for reasons concerning confidentiality, the client in the case study is referred to as Nigel at all times. This is a fictional name, which bears no resemblance to the client true name. Furthermore, any additional names, such as family members, have been changed to protect their identity. All details used within this report are factual and based on the psychological assessment conducted at the establishment the client attended.

CHAPTER 1

The Links Between Dynamic Risk Factors, Personality Disorder, and

Key Developmental Variables: A Narrative Review.

ABSTRACT

Sex offender risk assessments have developed considerably in recent years, particularly with the consideration of dynamic risk factors. These risk assessments consider a wide range of variables from static factors to personality and key developmental variables. There is a continuing debate as to the optimum risk assessment methods. Furthermore, these risk assessments provide indications for treatment and research into the links between risk factors and treatment adherence is ongoing. This review aims to discuss the links between dynamic risk factors, personality disorder, and key developmental variables, considering factors related to treatment adherence. Due to the breadth of the topics discussed in this review a narrative review with a systematic search was carried out. The review demonstrates that the research in this area positively demonstrates links between these variables and highlights the relevance to treatment. However, it is evident that further research in this area is pertinent in order to inform future risk prediction and treatment models.

INTRODUCTION

Collins and Fauser (2005) state that systematic reviews are best suitable for focussed topics and that narrative reviews are best suitable for comprehensive topics. The title for the current literature review is too broad to compile a systematic review or a meta-analysis. The review attempts to demonstrate the link between a number of topics which, until recently, have not been investigated as having relationships with one another. As such, the review comprises of a narrative review with a systematic search format.

Risk assessment of sexual offenders is a continuing topic within the empirical literature and is a topic of significant societal concern. Over the past few decades there have been significant advances in the field of sexual offender risk assessment, particularly with the research into dynamic risk factors, that is factors amenable to change over time (Bonta, 1996; Hanson & Harris, 2001). A number of risk assessments have been developed which consider dynamic variables and these have shown moderate predictive accuracy (e.g., Thornton, 2002). However, there is an ongoing need for further research into these variables and risk assessments. One particular issue in the risk assessment literature is that of the effects of personality on risk. Both psychopathy and personality disorders are considered in the empirical literature and included in a number of risk assessment measures (e.g., Hare, 1991; Hanson & Busierre, 1998). Furthermore, there has been interest in the role of key developmental variables and recidivism, particularly with reference to attachment theory (Hudson & Ward, 1997; Marshall, Hudson, & Hodkinson 1993; Marshall, 1993). Craissati (Craissati & Beech, 2001, 2004, 2006; Craissati, McClurg, &

Browne, 2002a, 2002b) has investigated this link and essentially concludes that there is evidence for the role of key developmental variables in the assessment of risk. Craissati and Beech (2006) propose a model to best predict risk by combining two or three key developmental categories (developmental trauma, childhood difficulties, insecure attachment) and static risk prediction. Again, there is a need for further research. However, these studies demonstrate the development of more comprehensive risk assessments. In addition, these risk factors and risk prediction models have been linked to treatment adherence, particularly personality and psychopathy (e.g., Craissati & Beech, 2004). However, empirical literature demonstrates mixed findings.

Aim of the current review

Currently, there are no reviews which attempt to link the relationships between key developmental variables, personality disorder and dynamic risk assessment. Moreover, no reviews investigate the relationship of these factors to the prediction of sexual offender treatment attrition. The areas of interest in the current review remain too broad to compile a Systematic Review or a Meta analysis. However, the author remained conscious of reducing the internal validity of the current review. Therefore, what is presented here is a narrative review using systematic search criteria, in order to investigate any existing relationships between personality disorder, key developmental variables, and dynamic risk assessments. In addition, the review aims to evaluate the use of these four key areas in relation to predicting treatment attrition in sexual offenders.

STRATEGIES EMPLOYED

Search of previous reviews

The Cochrane Library was extensively searched to highlight any previous literature reviews or Meta analyses associated with four key topics: risk assessment of sexual offending; personality disorder as a risk assessment for sexual offending; the role of key developmental variables; and treatment attrition of sexual offenders. In terms of risk assessment and sexual offending, the search revealed five :Clinical trialsø However, none of these reviews focused on which risk factors best predict sexual offending. In terms of personality disorder and sexual offending the results returned 31 -clinical trialsø, none if which were relevant. The role of key developmental variables in sexual offending returned no search results. With relevance to treatment attrition and sexual offenders, one Cochrane review, three in the reviews and two -clinical trials were revealed. However, only one clinical trial, one Cochrane review, and two other reviews were relevant to the search terms. Furthermore, there was no evidence of existing reviews examining the relationships between any of these areas. In addition, the search revealed no results for key developmental variables associated with sexual offending or personality/personality disorder. A list of the search results can be seen in Appendix 1.

Search Strategies

Initially, a scoping search was conducted in order to identify the relevant searches within the topic areas of interest. Due to the breadth of the current review this search was divided into four main topic areas, 1. Risk assessment measures, 2. Personality Disorder as a risk factor, 3. Key developmental variables, 4. Treatment adherence and

risk assessment (for details of the search terms see Appendix 2). The searches were restricted to English language publications and any comment papers and editorials were removed. The searches were manually scrutinised for any duplicate or untraceable papers. The results were then manually scanned by title and abstract to select articles relevant to the search terms. Reference lists were also manually searched for additional articles and any unpublished work or papers/posters presented at relevant conferences.

Access to publications

Publications were accessed through the University of Birmingham online library using Embase, Psychinfo, and Swetswise, and through direct contact with authors. A record of all search criteria can be found in Appendix 2. Relevant authors in the field were contacted including Dr. Jackie Craissati and Professor Anthony Beech. Both authors were able to make suggestions for further papers, which was useful as no search can be completely accurate and specific as they are limited to the search terms used. The search was also expanded after the date of the search terms based on consultation with my advisor Professor Anthony Beech.

This report will now examine each key area in more detail.

FINDINGS

Risk Assessment Measures

Actuarial risk measures

Risk assessment of sexual offenders has been topic of research within the literature for a number of years and as such there have been considerable developments in this area. Historically, clinical judgement was employed to determine the likelihood of an offender committing a similar offence in the future. However, this method has been criticised as highly unreliable and achieving reliability predictions of little more than chance. In response to this a number of authors have developed actuarial risk scales. These scales are based on static, unchangeable, risk factors such as negative family history, previous offending (sexual and violent), and relationship history. Typically, offenders are rated and categorised according to the predetermined risk levels on the associated measure. It is beyond the scope of this review to discuss the various actuarial risk assessments (e.g. Static-99) (Hanson & Thornton, 2000) (Risk Matrix-2000) (Thornton et al, 2003) and their predictive accuracy in detail. However, it can be noted that there is a considerable research base supporting the efficacy of actuarial risk measures (e.g., Allan, Dawson, & Allan, 2006; Kingston et al., 2008; Craig, Beech, & Browne, 2006) and they have been described as the most accurate method of risk prediction in sexual offenders (Beech, Fisher, & Thornton, 2003). Hanson, Broom, and Stephenson (2004) stated, using ROC AUC analysis, the most common actuarial measures all have moderate predictive accuracy for sexual offence recidivism (AUC = >0.7). Hanson and Morton-Bourgon (2004) argue that typically, the predictive accuracies of these measures have fallen within the moderate range

(AUC = > 0.7) and no single measure has shown to be superior across samples. Hanson and Bussière (1998) found that static variables best predicted long term recidivism in a review of 61 studies and comparative studies have consistently supported these assertions (e.g. Barbaree, Seto, Langton, & Peacock, 2001; Nunes, Firestone, Bradford, Greenberg, & Broom, 2002; Harris et al, 2003; Nunes, Firestone, Sjostedt & Langstrom, 2002).

Although actuarial measures have been described as the best tools available to assess the risk of sexual offending, they are not without their criticisms. In their paper Beech, Fisher and Thornton (2003) outline six important criticisms. 1. Actuarial scales do not indicate a certainty of whether an individual will offend. The risk of reoffending is described as a probability. 2. These probabilities of re-offending underestimate actual re-offence rates as the actuarial tools have been designed using official recidivism figures. 3. Actuarial measures do not consider individual differences and if used in isolation atypical variables may not be considered in the assessment of risk. 4. As these measures are developed based on a sample of sexual offenders, risk predictions for individuals presenting with any characteristics not presented within this sample may be inaccurate. 5. They omit acute risk factors, which may inform imminent recidivism risk as they seek to determine long term recidivism. 6. Perhaps most importantly, actuarial risk assessments do not offer anything in terms of which risk factors need to be managed to reduce the risk of recidivism.

Dynamic risk measures

In response to the criticisms of actuarial risk assessment, attention has been turned to dynamic risk factors. These are factors which are amenable to change over time (Bonta, 1996) and are stable, that is to say they are persistent over months or even years, or acute, that is they last for days or even only minutes (Hanson & Harris, 2001). To date many recidivism studies have been conducted, which identify the variables associated with sexual offence recidivism. Craig, Browne, and Stringer (2003a) identified a total of 24 dynamic variables, 10 stable dynamic (deviant sexual urges, sexual deviance, attitudes tolerant of sexual assault, cognitive distortions, lack of victim empathy, low self esteem, anger, substance abuse impulsivity, and personality disorder) and 14 acute dynamic (frequency of sexual fantasies, delinquent behaviour during treatment, deterioration in dynamic risk during treatment, poor treatment cooperation, deterioration in awareness of high risk situations, short duration of treatment programme, poor cooperation with supervision, isolation, unemployment, deviant social influences, chaotic lifestyle, poor social support, affective disorders, and substance abuse). Within the literature there is now a general consensus that deviant sexual interests and antisocial orientation are strongly related to sexual recidivism (Hanson & Bussière, 1998; Quinsey, Lalumiere, Rice & Harris, 1995; Hanson & Morton-Bourgon, 2005).

Bonta (1996) asserts that a risk-need approach is favourable, whereby risk is measured by static variables and need by dynamic variables. This argument was based on research into criminogenic needs (Andrews & Bonta, 1994; Andrews, Bonta, & Hoge, 1990) and the view that the likelihood of re-offending is reduced when these needs are altered (Bonta, 1996). Thus, asserting that criminogenic needs are essentially dynamic risk factors. Similarly, Beech, Fisher, and Thornton (2003) argue that an assessment of offender risk should contain information which can contribute to risk management rather than rely solely on simple documentation of risk. In

accordance with this, Beech, Friendship, Erikson, and Hanson (2002) examined the relationship between static and dynamic variables in a sample of child abusers and found the use of dynamic variables significantly increased the predictive accuracy of the Static-99. Reviewing the literature, Hanson, Broom, and Stephenson (2004) asserts that although there are currently no well-established dynamic risk prediction variables, they are clearly too influential to be ignored.

From such research findings there are three notable risk assessments which consider dynamic variables, the Initial Deviance Assessment (IDA) (Thornton, 2000, 2002), The STEP test battery (Beech, Fisher, & Beckett, 1999), and the Stable-2007 (formally Stable-2000 and Sex Offender Needs Assessment Rating) (Hanson & Harris, 2001; Hanson, Harris, Scott, & Helmus, 2007).

Initial Deviance Assessment (IDA) (Thornton, 2000, 2002)

Thornton (2002) supports the idea of a Structured Risk Assessment (SRA). Such an assessment seeks to integrate static risk assessment (Static-99), stable dynamic risk assessment (using the IDA), progress evaluation based on response to treatment, and management of risk based on acute dynamic factors and offence specification. The IDA incorporates four domains of dynamic risk predictors: Sexual Interests, Distorted Attitudes, Socio-effective Functioning, and Self Management. The IDA is empirically informed as the four domains are well established within the empirical literature spanning three different research designs; longitudinal (e.g., Bakker, Hudson, Wales, & Riley, 1998; Hanson & Bussière, 1998), group comparisons (e.g., Barbaree & Marshall, 1989; Beech, Fisher, & Beckett, 1999), and offence precursor studies (e.g., Hanson & Harris, 2001; Pithers, Kashima, Cumming, & Beal, 1988; Proulx, Perrault,

& Ouimet, 1999; Ward, Louden, Hudson, & Marshall, 1995). Thornton (2002) posits that this research base provides enough evidence for the justification of these risk factors. Moreover, Thornton (2002) reports that the predictive accuracy of the IDA is comparable to Static assessments with ROC coefficients in the 0.7 range. However, the Sexual Interests domain was not included in these studies, which leaves it unvalidated for use in the IDA. Further, the IDA has received criticism for not considering response to supervision as a dynamic risk factor (Lindsay, Elliot, & Astell, 2004).

The STEP test battery (Beech, Fisher, Becket, 1999)

The STEP test battery is used nationwide by the Prison and Probation Services in the U.K., following its successful implementation as an evaluative tool in the Prison Sex Offender Treatment Programme Project. This assessment is a psychometric test battery designed to estimate dynamic risk factors (pro-offending attitudes, socio-affective problems, and deviant sexual interest) in adult male child abusers. Beech, et al. (1999) advocate that the presence and intensity of these dynamic risk factors can be used to categorised offenders into high and low deviancy. Beech, Friendship, Erikson, and Hanson (2002) found that establishing the deviancy level of offenders pre-treatment allows the predictive accuracy of the Static-99 to improve by between 25% and 86%. However, the STEP-test battery has its limitations in that it does not currently have the capacity to generalise to other populations as it has only been validated for use on adult male child abusers. As such, Craissati and Beech (2003) highlight the requirement for ongoing research into the applicability of the STEP test battery to other client groups.

The Stable-2007 is a 13 item (Significant Social Influences, Capacity for Relationship Stability, Emotional Identification with Children, Hostility toward women, General Social Rejection, Lack of concern for others, Impulsive, Poor Problem Solving Skills, Negative Emotionality, Sex Drive/Sex Preoccupation, Sex as Coping, Deviant Sexual Preference, Co-operation with Supervision) stable dynamic risk assessment. The Stable-2007 can be used with both adult rapists and child sexual abusers. However, the item Emotional Identification with Children is only rated in child sexual abusers. Each of the items in this assessment are rated 0, 1, or 2 for not present, maybe present, and definitely present respectively. These are then tallied to reach a total Stable-2007 score where 0-3 is low risk, 4-11 moderate risk, and 12+ high risk. Hanson, Harris, Scott, & Helmus, (2007) report that the results of their study investigating the predictive accuracy of the Stable-2007 are consistent with previous studies demonstrating that stable dynamic characteristics significantly contributed to the prediction of recidivism above that given by established static measures. Furthermore, Hanson, et al. (2007) argue that in order to produce a holistic risk assessment the Static-99 and Stable-2007 should be combined with their acute dynamic risk assessment the Acute-2007, which produces an overall risk score whilst informing on which variables should be targeted during treatment. However, further work needs to be conducted into the predictive accuracy of this assessment as it remains in the investigative stages at present and no concrete conclusions have been drawn.

Summary

It can be seen that the process of risk assessment has evolved to a position whereby the predictive accuracy of static variables associated with recidivism are well established and continuing research is addressing the development of the dynamic variables. Indeed, common themes and generally accepted stable dynamic risk predictors can be distinguished in the literature. Nonetheless, there is a ubiquitous need for the development of comprehensive risk assessment tools, which comprise well-validated static, stable dynamic, and acute dynamic factors which can be achieved via further research into sexual recidivism. Indeed, Bonta (1996) states that advances in the assessment of offenders are continuing at an exponential rate and no doubt there will be a fourth generation in risk assessment in the near future.

Personality Disorder has been identified as a likely risk factor for both sexual and violent offending. This will be discussed in more detail in the next section.

Personality Disorder and Its Relationship to Risk

The most widely used definition of personality disorder (PD) is from the American Psychiatric Association (APA), Diagnostic and Statistically Manual of Mental Disorders-Fourth Edition-Text Revision (DSM-IV-TR, APA, 2000). This states that personality disorder is :an enduring pattern of inner experience and behaviour that deviates markedly from the expectations of the culture of the individual who exhibits it DSM-IV-TR stipulates that in order to receive a formal diagnosis of personality disorder the pattern must be manifested at least two of the following areas: cognition, affect, interpersonal functioning, and impulse control. The enduring pattern needs to be inflexible, pervasive across a range of personal and social situations, lead to

clinically significant distress, be stable over time, and not related to the effects of a substance use or other medical condition. The onset of personality disorder is usually marked in adolescence or early adulthood.

The most popular assessment of personality disorder other than DSM-IV is the Millon Clinical Multiaxial Inventory (MCMI-III). The MCMI-II uses items reflective of the diagnostic criteria of the DSM to diagnoses personality disorders and Axis 1 disorders; moreover, it is the most comparable to the DSM-IV personality disorder constructs. However, it does not assess other mental health problems. Within the MCMI-III there are 175 items relating to 24 clinical scales and three modifier scales (Disclosure, Desirability, and Debasement). The 24 clinical scales are comprised of 11 Clinical Personality Patterns (Schizoid, Avoidant, Depressive, Dependant, Histrionic, Narcissistic, Antisocial, Sadistic, Compulsive, Negativistic, and Masochistic), three Severe Personality Pathologies (Schizotypal, Borderline, and Paranoid), and ten Axis 1 and other clinical syndromes (Anxiety, Somatoform, Bipolar, Dysthymic disorder, Alcohol dependence, Drug dependence, Posttraumatic Stress, Thought Disorder, Major Depression, and Delusional Disorder). Scores above 75 on any scale of the MCMI-III indicate the probable presence of a disorder and scores over 85 signify definite presence of pathology. The MCMI-III is relatively easy to administer and to score. However, it is only validated on clinical populations and is susceptible to bias during interpretation.

Other personality assessments include the Minnesota Multiphasic Personality Inventory-Second Edition (MMPI-II) (Butcher & Megargee, 1989, 2001) and the

Personality Assessment Inventory (PAI) (Morey, 1991). In comparison to the MCMI-III, the MMPI-2, at 567 items is a far longer and timely assessment. However, it is a widely researched, which also has credible measures of validity. The PAI is also lengthier than the MCMI-III although approximately 40% shorter than the MMPI-2. An advantage of the PAI is that it has three normative samples (community, psychiatric, and college), allowing for a broad source of comparison.

DSM-IV records ten personality disorders which can be presented in three clusters: Cluster A (odd or eccentric disorders) includes: Paranoid, characterised by irrational distrust and suspiciousness of others believing that others motives are malicious; Schizoid, characterised by a limited range of emotional expression and a lack of interest in social relationships; Schizotypal, characterised by eccentric behaviour and a reduced capacity for and severe discomfort with close personal relationships. Cluster B includes: Antisocial, characterised by manipulative behaviours, difficulty controlling impulses with a disregard for others rights, and a lack of empathy; Borderline, characterised by an inability to regulate emotion resulting in dramatic and abrupt shifts in mood, impulsivity, poor self-image and tumultuous interpersonal relationships; Histrionic, characterised by melodramatic behaviour, attention seeking, and an excessive level of emotionality; Narcissistic, characterised by grandiosity, need for admiration, and lack of empathy. Narcissistic individuals tend to be selfabsorbed, intolerant of othersø perspectives, insensitive to othersø needs and indifferent to the effect of their own egocentric behaviour. Cluster C includes: Obsessive-Compulsive, characterised by a preoccupation with perfectionism and orderliness. There is a lack of flexibility, which interferes with the ability to complete tasks effectively and damages personal relationships; Avoidant, characterised by an

intense level of social anxiety, feelings of inadequacy, extreme self-consciousness, hypersensitivity to rejection, and a subsequent avoidance of social situations; characterised by neediness of others, submissive behaviours, over sensitivity to criticism, and extreme fear of separation; and Dependent, characterised by a severe psychological dependence on other people, feelings of helplessness and submissiveness, and a need for constant reassurance.

In addition, there are a number of personality disorders which have been removed from the DSM-IV such as passive-aggressive, depressive, or sadistic. However, these can still be conceptualised under the term personality disorder not otherwise specified, which can be diagnosed under the criterion that the individual displays symptoms of two or more personality disorders and this impairs their social and interpersonal functioning. Furthermore, Psychopathy is considered to be a serious personality disorder. Although not a personality disorder diagnosed within DSM-IV, it is often associated with antisocial personality disorder (Hart, Forth, & Hare, 1991). However, this association has caused confusion among the diagnoses (Hare, 1998) and in fact, most adults diagnosed with antisocial personality disorder do not fulfil the criteria for psychopathy as they lack the callous and emotionless features (Hart & Hare, 1997).

Cleckley (1982) was one of the first researchers to define psychopathy, stating that psychopathic individuals exhibit chronic maladjustment, superficial personal relationships, are highly manipulative, and have an extreme lack of empathy. In terms of criminal behaviour, Cleckley (1941) postulated that this is a result of impulsivity combined with the inability to form affective bonds with others. Drawing on Cleckley (1941) earlier work, Hare (1991) developed the most widely known model

of psychopathy, whereby psychopathy is characterised by a combination of affective, interpersonal, and behavioural characteristics, which have serious and devastating social consequences (Hare, 1993, 1996). These features are evident from adolescence or early adulthood, are relatively stable over time, and appear to be hereditary (Lynam, Caspi, Moffitt, Loeber, & Stouthamer-Loeber, 2007; Viding, Frick, & Plomin, 2007). Moreover, Viding, Blair, Moffitt, and Plomin (2005) have evidenced that callous-unemotional traits observed in seven year olds have a strong genetic influence. Other researchers have reported environmental influences, such as abuse in childhood (Weilder & Widom, 1996) and poor family functioning (Farrington, 2006).

To measure psychopathy, Hare (1991) developed the Psychopathy Checklist-Revised (PCL-R) (1991), which is now the most widely used assessment tool for this personality construct. The PCL-R consists of 20 items each scored on a three point scale by a qualified assessor via file information and semi-structured clinical interview. The total score is out of 40 and typically scores of 25 or above for U.K. samples and 30 or above for U.S. samples are considered to be psychopathic. The PCL-R was developed in such a way that an individual@ score (ranging from 0-40) provides an estimation of the degree to which the given individual matches the prototypical psychopath. Initially, the PCL-R listed two factors, Factor 1 assessing the interpersonal and affective characteristics of psychopathy and Factor 2 assessing the chronic antisocial behaviour. This model is well published (e.g. Hare, 1996a, 1996b) and has been cross-validated over a number of diverse samples (Cooke, 1995). However, it did not fare well across African American and Caucasian populations (Kosson, Smith, & Newman, 1990) or substance abusing populations (McDermott et al., 2000). More recently, a four factor model has been proposed (Newman, Kossan,

Hare, & Forth, 2006) and has received some empirical support (e.g. Hill, Newman, & Rogers, 2004). There is still continuing debate in the literature whether a one, four, or even three factor model is superior and clearly more research needs to be completed in this area. However, it is beyond the scope of this review to discuss these issues further.

Prevalence in sex offenders

The prevalence of both personality disorder and psychopathy in sexual offenders has been documented within the extant literature. In terms of personality disorders, there is no direct association between personality disorder and sexual offending (Craissati, Webb, & Keen, 2005). However, studies have reported a high prevalence of personality disorder in sex offenders. For example, Borchard, Gnoth, and Schulz (2003) reported at least 72% of their sample met the criteria for at least one personality disorder, with the highest prevalence being in Cluster B disorders and mainly antisocial personality disorder. Berger, Berner, Bolterauer, Gutierrez, and Berger (1999) found that 89% of the sample exhibited at least one personality disorder. Further, Chantry and Craig (1994) found that, using the Millon Clinical Multi-axial Inventory-III (MCMI-III) (Millon, Millon, Davis, & Grossman, 1997) adult rapists demonstrated either an emotionally detached personality with dependent personality features, or an independent personality style characterised by narcissism and antisocial features, whilst child sex offenders demonstrated a primarily detached personality style, with dependent personality traits, with or without passive-aggressive features. In terms of psychopathy, Borchard, Gnoth, and Schulz (2003), using a cutoff of 25 on the PCL-R found that 21% of their sample were considered psychopathic.

Similarly, Forth and Kroner (1995) found that 26% of rapists, 18% of rapist/child abusers, and 5% of incest offenders reached the cut-off for psychopathy.

Personality Disorder, Psychopathy, and recidivism

The research in relation to personality variables and recidivism has focused on a criminal personality involving antisocial personality disorder and/or psychopathy as a risk factor for further offending. The evidence to suggest that a range of personality disorders/variables are at play in increasing risk is relatively sparse. Harris, Rice, and Cormier (2002) have reported a history of personality disorder to be related to recidivism. Craissati, Webb, and Keen (2005), in an attempt to devise an enhanced model of risk prediction, demonstrated that a screening tool which included, recidivism risk (measured by static or dynamic risk measures), the presence of childhood trauma, and mental health difficulties or personality disorder was the strongest recidivism prediction model. However, these findings do not define between diagnosis of mental illness and different types of personality disorder.

There is a considerable amount of research demonstrating correlations between APD, psychopathy, and recidivism. Hanson (2000) demonstrated that antisocial personality in a group of sexual offenders achieved correlation co-efficients of r=.16 for general recidivism and r=.14 for sexual recidivism. Similarly, Hanson and Bussierre (1998) in their meta-analytic review reported correlation co-efficients of r=.14 for antisocial personality and sexual recidivism (correlation co-efficients are a measure of the proportion of variance shared by two variables. This ranges from 0 to 1, where 1 is a perfect positive linear relationship). Langstrom, Sjostedt, and Grann (2004) reported antisocial personality as a risk factor for recidivism. In terms of psychopathy, a

number of studies have found psychopathy and sexual recidivism to have a significant relationship (e.g., Hanson & Harris, 1998, 2000; Hildebrand, de Ruiter & de Vogel, 2004; Quinsey, Rice, & Harris, 1995; Rice, Harris, & Quinsey, 1990). Moreover, in their meta-analysis Hanson and Morton-Bourgon (2004) found significant overall Effect Sizes for psychopathy in predicting sex offender recidivism (sexual recidivism r =.15). However, other studies have not replicated these findings (Barbaree, Seto, Langton, & Peacock, 2001; Langstrom & Grann, 2000; Sjostedt & Langstrom, 2002).

Clearly more research needs to be conducted in this area. However, there has been investigation into sexual deviance as a mediator for psychopathy and sexual recidivism. Rice and Harris (1997) and Harris et al. (2003) found that sex offenders scoring over 25 on the PCL-R and were assessed as having a deviant sexual preference, had higher and faster rates of both sexual and violent recidivism. In addition, some discrepancy in these research findings may be as a result of the differences in the way researchers operationalise recidivism (Wormith, Oliver, Stevenson, & Girard, 2007). Wormith et al. (2007) suggest that there may be differences in the way researchers define recidivism, such as arrest, re-conviction, resentence as well as on dimensions such as the type of offending behaviour classed as recidivism. Moreover, the length of follow up of recidivism studies significantly impacts on the recidivism rates, particularly with sexual recidivism where the base rate of offending is typically very low.

The next section aims to review what is known about the role of developmental variables and their relationship to sexual offending and risk assessment.

Developmental Variables

There has been interest in attachment theory and its role in sexual offending (Hudson & Ward, 1997; Marshall, Hudson, & Hodkinson 1993; Marshall, 1993). Attachment theory posits that based on early childhood experiences, individuals form working models of themselves, others, and the world. It suggests that these experiences are primarily determined by significant others, which then determine how attachments are formed with others in later life (Bowlby, 1977; 1979). Moreover, abnormal adult emotional bonds are seen as a result of distortions in early attachments. Marshall, et al. (1993) suggested that individuals who experience poor attachments in childhood develop low self esteem, which results in difficulties in forming emotional intimacy. As such, these individuals may be more likely to seek this intimacy through sexual experiences, which may not be consensual. Further, both sexual offending and poor attachments have been linked to the tendency to sexually objectify others and to experience a lack of empathy (Marshall, et al., 1993).

The empirical evidence demonstrating the relationship between sexual offending and early attachment experiences is relatively sparse despite a wealth of theoretical, clinical, and anecdotal support (Craissati, McClurg, & Browne, 2002). However, Hanson and Bussière (1998) reported that negative relationship with mother was the single developmental history variable predictive of sexual offence recidivism. It is important to highlight the relevance of this finding whilst considering that a small sample size and only three studies evidenced this. Further, insecure attachments styles have been observed in sexual offenders. Smallbone and Dadds (1998) found that in comparison to property offenders, sexual offenders were less secure. Ward, Hudson, and Marshall (1996) found that child molesters exhibited a fearful/preoccupied

attachment style, whereas rapists demonstrated a dismissive style. In addition, more specific key developmental variables associated with attachments in childhood have been studied, although this link is yet to be clearly established (Craissati & Beech, 2006). There are consistent findings within the evidence base that sex offendersø backgrounds are marked by neglect, violence and disruption (Bard et al., 1987; Craissati & McClurg, 1996). Studies have primarily demonstrated that maltreatment in childhood and in particular sexual victimisation are related to later sexual offending. Craissati, Falla, McClurg, and Beech (1998) found that risk was associated with a history of childhood victimisation and some prospective studies have illustrated that sexual or violent offences are committed by one in six maltreated children (Watkins & Bentovim, 1992; Widom; 1989).

In a series of papers Craissati (Craissati & Beech, 2001, 2004, 2006, Craissati, McClurg, & Browne, 2002a, 2002b) has aimed to cement the research on the relationship between key developmental variables and sexual offending. This has been achieved largely through the research on the Challenge Project, a community assessment and treatment programme for sexual offenders. Craissati and Beech (2001) focussed on attrition of sexual offenders. They found that the following variables: \(\pm\) two or more childhood difficulties\(\ph\) and \(\pm\)childhood sexual victimisation\(\ph\) were significantly associated with both non-compliance and attrition in treatment. However, they noted that it was difficult to obtain the background variables of the participants and strict criteria for treatment attendance were employed, i.e. failure to attend the complete programme for any reason was deemed as attrition. Therefore, the remaining sample size was reduced. Craissati, McClurg, and Browne (2002a) investigated the effects of sexual victimisation in childhood. In this study almost half

of the sexual offenders had been sexually victimised in childhood, most often by an acquaintance or stranger. Only 18% of the sample reported no physical, sexual, or emotional abuse, lending weight to the link between key developmental variables and sexual offending. Furthermore, those who were sexually victimised as children reported higher levels of childhood difficulties such as being bullied and having difficulties in friendships, resulting in adult characteristics such as self harm, hostility, lack of empathy, sexual pre-occupation, and contact with mental health services. However, this study relied heavily on the self report of the offenders and only included child sexual offenders, therefore, no conclusions can be drawn for rapists.

Craissati, McClurg, and Browne (2002b) studied both child sexual offenders and rapists, primarily investigating the parental bonding experiences of these individuals. Using the parental bonding instrument (PBI) they found that child sexual offenders and rapists did not differ on the four categories (affectionless control, affectionate constraint, weak bonding, and optimum bonding). However, in terms of maternal and paternal care the child sexual offenders scored significantly higher than the general population, whereas rapists did not. In addition, they reported that both groups reported high levels of affectionless control which is categorised by inconsistent parenting and related to insecure attachments. Moreover, poor parental bonding was linked to adverse childhood experiences, which was particularly evident for child sexual offenders. However, it is notable that the sample size of rapists was particularly small (19 rapists compared to 57 child sexual abusers). Craissati and Beech (2004) acknowledge that the research on adult rapists is relatively sparse and the empirical evidence demonstrates mixed findings. They compared child sexual offenders and rapists on a range of background and offence related variables and

found little variation between the groups. Rapists were, however, less likely to have been sexually victimised in childhood and demonstrated more difficulties with intimacy. Although, rapists who had been sexual victimised showed greater levels of psychosexual disturbance than those who were not.

Craissati and Beech (2006) attempted to produce a model which would integrate the key findings of the previous research on the Challenge Project. They suggest that the evidence is there to support the notion that key developmental variables are related to both risk assessments and treatment compliance/attrition. They suggest a model that includes a combination of two or three key developmental categories (developmental trauma, childhood difficulties, insecure attachment) and static risk prediction as the optimum method of determining risk of community failure. Although, there is clearly a need for further research on key developmental variables and later sexual offending there is clearly a link between these factors. Furthermore, the existing research gives valid suggestions for treatment implications following this model, which will be discussed further.

In the next section, treatment adherence and its relationship to risk assessment will be examined in more detail.

Treatment adherence and Risk Assessment

Existing treatments

In England, Canada, the U.S., and New Zealand; cognitive behavioural interventions are largely utilised for sex offender treatment in both the Prison Service and community settings (Craig, & Beech, 2009). Psychodynamic approaches are the basis

for intervention with this population in the UK NHS. Cognitive behavioural interventions include a range of treatments stemming from social learning theory and essentially strive to modify the individuals cognitive and emotional functioning and their behaviour. This involves linking thoughts, feelings, and behaviours with respect to a presenting difficulty (e.g., sexual offending, sexual fantasy, etc.) and a provision of alternative strategies to cope with the presenting difficulty. In addition, in sex offender treatment, cognitive behavioural therapy will usually include a relapse prevention module. Psychodynamic interventions can have many definitions, which are largely inefficient (Kenworthy, Adams, Brooks-Gordon, & Fenton, 2004). A well defined psychodynamic approach can be described as providing individual sessions, which are regular, are at least 30 minutes in duration, and continue for at least one year. These sessions would be provided by a trained psychoanalyst and they need to be working where they target sexual relations at the infantile level. Further, Kenworthy et al., (2004) stated that sessions rely on a variety of strategies, including explorative insight-orientated, supportive or directive activity, applied flexibly. Other less common but still readily reported therapies include drug treatment such as Medroxyprogesterone, which increases the level of female hormones in the body to reduce the sex drive of the male sexual offender and behavioural interventions which may include; electrical aversion, relaxation treatment, self regulation, imaginal desensitisation, and covert sensitisation.

Methodological Difficulties

 intervention were to be theoretically sound, the use of an RCT brings controversy in terms of ethics, human rights, and the possible repercussions of potential victims given that a beneficial treatment would essentially be withheld (Hood, 2002). Furthermore, historically, positive treatments effects in quasi-experimental treatment programmes have been sparse (Quinsey, 1998). Failure to complete treatmentø was found to be a moderate predictor of sexual recidivism in one meta-analysis. However, this may have been more likely a result of risk monitoring than the outcome of the treatment programme (Hanson, 2002). One of the most pertinent issues in evaluating sex offender treatment programmes is how the treatment is measured. Using sex offendersø self-report or the reports of others including professionals can often be unreliable. The use of recidivism rates can be used to assess treatment success. However, with the known difficulties of underreporting and under-recording of offences coupled with the fact that even of those offenders arrested few are convicted this measurement method also presents as problematic (Grossman, Martis, & Fichtner, 1999). In addition, the base rates for recidivism of sexual offenders are typically low, approximately 15% after five years and 20% after ten years (Hanson & Bussière, 1998; Hanson & Thornton, 2000). As such, the study design may become corrupted over time when such long follow up periods are required.

Further, many of the sex offender treatment programmes were designed with the child sex offender in mind and as such rapists are under-represented in many of the evaluation studies (Ward, et al. 1997). This may be attributed to a number of factors such as higher denial rates of offending behaviour (Kennedy & Grubin, 1992) or rapists being more likely to drop out of treatment, more likely to recidivate, and be more antiauthoritarian (Craissati & Beech, 2004). Therefore, questions arise as to the

utility of these programmes with rapists. Treatment efficacy studies produce varying results when considering rapists. Marques, Day, Nelson, and West (1994) documented significant differences between treated and untreated rapists (9.1% and 27.8% respectively). However, this study only included 26 rapists of which 11 dropped-out before the treatment end date. Conversely, Alexander (1999) reported non-significant treatment differences, where 20.1% of the treatment group and 23.7% of the non-treatment group sexually re-offended.

Despite the above, a number of authors have managed to compile meta-analyses of sexual offender treatment efficacy. For example, Hall (1995) reported a small treatment effect when considering studies which compared a treatment group to a comparison group. This analysis also demonstrated cognitive behavioural therapy to be the most effective. However, as the comparisons made in the included studies were largely between individuals who completed treatment and those who did not, this may reflect a difference in recidivism rates due to individual differences rather than treatment effects (Hanson, Broom, & Stephenson, 2004). Hanson et al. (2002) report that studies comparing treatment completers to drop-outs evidence higher recidivism rates in drop-outs regardless of treatment type. They concluded that these results reflected a methodological flaw rather than a real treatment effect. Hanson et al. (2002) also made comparisons using studies based on matching/incidental principles and a number of RCTøs. They reported that cognitive behavioural treatments produced significant reductions in sexual offence recidivism. However, behavioural approaches and psychotherapy yielded no treatment effects. Schmucker and Losel (2008) compiled a meta-analysis, using 69 studies with a total of 9,512 treated sexual offenders and 12,669 untreated offenders. A positive treatment effect was

demonstrated with cognitive behavioural treatments being reported to be superior to other approaches.

More recently, Hanson, Bourgon, Helmus, and Hodgson (2009) completed a metaanalysis of 22 studies to investigate the effectiveness of sexual offender treatment (3,121 treated offenders, 3,625 comparison group offenders). In particular they examined whether the risk-need-responsivity principles (Andrews, Bonta, & Hoge, 1990; Bonta & Andrews, 2007) were applicable to the treatment of sexual offenders. Further, they considered whether the meta-analysis results would be confounded by study design, selective publication of positive results, age of sample, and treatment setting. They demonstrated a positive treatment effect, with the rate of sexual recidivism being lower for treated offenders. Hanson et al. (2009) also reported smaller, albeit non-significant, treatment effects in studies they considered to be of good quality than weak studies. Age of the offenders was not a mediating factor of treatment adherence, nor was treatment setting, that is community versus institutional treatment. Moreover, they report that those studies which represented programmes adhering to the risk-need-responsivity principle returned stronger treatment effects. These treatment effects were increased dependant on the number of principles adhered to. The risk-need-responsivity principle refers to three principles which if adhered to are generally accepted to increase intervention success. The risk principle stipulates that those offenders falling in the higher risk categories should be receiving the most intense treatment programmes and low risk offenders should receive much lower intensity or no treatment. The need principle states that criminogenic needs i.e. factors which are related to recidivism should be the primary treatment targets. The responsivity principle refers to adapting treatment programmes to the individual on

factors such as motivation, readiness to change, personality style etc (e.g., Andrews & Bonta, 2003; Andrews, Bonta, & Wormith, 2006; Andrews & Dowden, 2006; Andrews et al., 1990; Gendreau & Andrews, 1990). However, it is important to note it is not without its criticisms. Primarily in that it can be argued that is alone does not provide enough therapeutic tools to work with offenders fully (Ward & Gannon, 2006). Although this model has a strong empirical base and a clear direction for treatment, it fails to consider the over all welfare of the individual. Furthermore, as Mann, et al. (2004) indicate, goals which focus on reducing offending are more motivating for the individual than approach related goals i.e. those goal which focus on pursing a better life. Moreover, the evidence base relies heavily on meta-analytic studies which in themselves can be flawed due to which studies are included and which perhaps unpublished studies may be excluded i.e. the file draw phenomenon (it is much harder to get studies published that support the null hypothesis than those which report positive findings).

Hanson et al (2009) discuss that although positive treatment effects were reported, the confidence in these findings are reduced due to weak study designs, which further highlights the need for more rigorous research in this area. However, it is also suggested that more recent studies produced stronger treatment effects, which may be reflective of better treatment programmes and or better research designs.

Further, the Correctional Service of Canada reported on their programme (Community Sex Offender Program, CSOP) that offenders released after the program and those released before demonstrated few differences in recidivism rates (CS/RESORS). The approach with the most promising results was cognitive

behavioural therapy, which mainly focussed on offence specific variables (Stephenson, 1991). Hanson, Broom, and Stephenson (2004) extended the follow up of this study to 12 years and report that no significant treatment effects were observed. However, more recently, Olver, Wong, and Nicholaichuk (2009) demonstrated that, using a treatment programme adhering to the :What Worksø principles. What Works is a set of principles outlined by National Offender Management System (NOMS) which outline what works i.e. effective treatment and management of offenders and what does not work in reducing recidivism. The What Works literature produces guidelines, based on best practice for working with offenders and from this a number of standarised treatment programmes have been developed such as Enhanced Thinking Skills, Reasoning and Rehabilitation, and the Sex Offender Treatment Programme (Harper & Chitty, 2005). Over a twenty year follow up treated offenders recidivated significantly less than the non-treated comparison group. They argue that studies published in recent years are methodically sounder than those included in earlier reviews. In addition, they posit that the positive treatment effects observed in the evaluation of their programme (Clearwater Program, a high-intensity inpatient sex offender treatment program) were largely due to the cognitive behavioural nature of the programme, that they adhere to the principles of risk, need and responsivity, and that there were high enough base rates to detect these treatment effects (Barbaree, 1997). In addition, Crassati and Beech (2001) report on the Challenge Project that when controlling for risk level, significant differences in exhibiting :sexually risky behavioursø were found between treated and untreated high risk offenders, 14% and 19% respectively. We can conclude that historically, the results of treatment for sex offenders have been poor. However, with more rigid cognitive-behavioural based programmes in place and improve methodology of these

studies we should expect better results in the future. Furthermore, given the importance of sex offender treatment even small treatment effects warrant a continuation of a programme due to the impact on society and the victims.

In addition, it has been suggested that there may be variation in terms of the types of offenders studied in regards to treatment effectiveness. Seto and Barbaree (1999) in their study reported no differences between types of sexual offenders (rapists, incest offenders, and extra familial child abusers) on their scores on treatment behaviour. Craissati and Beech (2004) reported that child molesters performed better than rapists in terms of treatment attendance and overall compliance. However, this study only included a small number of rapists, which may affect the findings. In terms of recidivism, rapists were more likely to commit any new offence than child abusers but no differences were found between the groups when considering new serious offences (a new violent or sexual offence).

Psychopathy, Personality, and Treatment

Seto and Barbaree (1999) in their study reported no differences between types of sexual offenders (rapists, incest offenders, and extra familial child abusers) on their scores on treatment behaviour. In terms of recidivism, rapists were more likely to commit any new offence than child abusers but no differences were found between the groups when considering new serious offences (a new violent or sexual offence). Furthermore, Both the PCL-R and treatment behaviour have been shown to be significantly associated with serious recidivism. Those who scored highly on the PCL-R combined with demonstrating good treatment behaviour were more likely to commit any new offence and five times more likely to commit a new serious offence.

In addition, Keihl, Hare, McDonald, and Brink (1999) found high factor 1 scores on the PCL-R to be predictive of recidivism following a sex offender treatment programme. Neumann, Kossan, Forth, and Hare (2006) postulates that individuals with high Factor 1 scores may in fact learn further manipulative skills during treatment which leads them to be at greater risk of serious re-offending in the future, which must be taken as a serious consideration when assessing psychopathic sexual offenders for relevant treatment programmes. Conversely, DøSilva, Duggan, and McCarthy (2004) and Barbaree, Langton, and Peacock (2006) both found that rates of sexual recidivism in psychopaths does not increase after treatment. Doren and Yates (2008) in their review of the empirical literature concluded that no firm conclusions of the amount of benefit psychopathos gain from treatment can be drawn from the current literature. They assert that some psychopaths did not demonstrate the same recidivism rates as non psychopaths, whereas other did. Furthermore, there were no clear differentiating characteristics between those psychopaths who benefited from treatment and those who did not. As such, this is a clear topic for further research. In the mean time unfortunately we must employ the current literature as best we can and use clinical judgement to determine the likelihood of a psychopathic sex offender sex risk increasing or decreasing through treatment.

In terms of developmental variables, Craissati, McClurg, and Browne (2002a) report that, due to the difficulties sexual offenders experience in terms of parental bonding, offenders may benefit from increased victim empathy work. They suggest that this could be achieved by increasing the offendersø awareness of the style of their attachments and parental relationships as well as their own victim experiences. This would assist in facilitating changing dysfunctional ways that they relate to others.

Craissati, McClurg and Browne (2002b) suggest that an offender who has been sexually victimised themselves in childhood may struggle to form appropriate victim empathy and as such they are at risk of offending. They suggest that hearing the offender as a victim during treatment may lead to improved treatment results. Personality disorder and its relationship to risk of recidivism and treatment noncompliance have been discussed in the previous sections of this review. Although it has been consistently demonstrated that personality disorder is a risk factor for further offending, there is little evidence on what works for personality disordered sex offenders. Craissati, Webb, and Keen (2005) have reported on their programme for personality disordered offenders and have suggested that specialist interventions such as the Challenge Project are the best way to manage this high risk group. They demonstrated higher levels of mental health difficulties, more childhood victimisation and more negative developmental variables. These individuals were also more likely to fail the community treatment programme. As such, they suggest there is a need for a specialist programme designed specifically to engage offenders who are at risk of failure, which is certainly an area for future research. Moreover, given the mixed findings in terms of treatment success depending on type of offender, personality disorder, and development variables, further research into the development of more specialist programmes is warranted in the future.

Summary and Conclusions

The area of risk assessment has grown considerably in recent years, particularly in relation to which factors best predict recidivism (Hanson & Morton-Bourgon, 2004). Actuarial risk assessments, based on static factors, have been developed and have been found to have relatively high predictive accuracy (e.g. Hanson, 2004, AUC >.7).

However, they fail to consider risk management issues and as such dynamic variables have been identified (e.g. Beech, Fisher & Beckett, 1999; Hanson & Harris, 2001). In addition, personality disorder has been identified as a likely risk factor for recidivism and is considered to be so in a number of risk assessment tools (e.g. Hare, 1991; Webster, Eaves, Douglas, Winthrup, 1995). However, the research in this area needs to be extended.

More recently key developmental variables have been associated with sex offender recidivism risk (e.g. Craissati, McClurg, & Browne, 2002b). Craissati (Craissati & Beech, 2001, 2004, 2006, Craissati, McClurg, & Browne, 2002a, 2002b) also highlighted a possible link between personality disorder and these key developmental variables and as such questions whether these are in fact measuring the same or a similar construct. This research has demonstrated that this is a valid hypothesis, however, it remains clear that further research needs to be conducted to explore this as there is currently limited supportive evidence in sexual offenders. However, there is a plethora of research in the general personality disorder research demonstrating the link to negative childhood experiences. For example, Fonagy (2000) suggests that Borderline Personality Disorder characteristics may be grounded in developmental pathology. Weaver and Clum (1993) reported that individuals with a diagnosis of Borderline Personality Disorder reported significantly more history of sexual abuse, physical abuse, and witnessing violence than those who did not have this diagnosis. Laporte and Guttman (1996) indicate that women with Borderline Personality Disorder reported more separation, abuse, and loss than individuals with other personality disorder diagnoses. Further, the development of both psychopathy and Anti-social Personality Disorder has been related to a history of attachment problems.

Indeed, psychopathic traits can be demonstrated form the age of eight and as such it has been perceived as a developmental disorder (Blair, 2005; Blair et al., 2006). Battle et al., (2004) assessed history of abuse and negative childhood experiences in 600 individuals with a diagnosis of either Borderline, Schizotypal, Avoidant, or Obsessive-compulsive personality disorder and found that 73% self reported abuse and 82% self report neglect. Furthermore, there may well be many pathways to the same personality disorder i.e. different risk factors or traumas may result in the same diagnosis in adulthood. Moreover, there will be many risks associated with a disorder in an individual as findings rarely demonstrate a single variable to be associated with a disorder (Rutter, 1987; 1989).

However, it is important to note that these associations between trauma and personality disorder are subject to problems in base rate due to many studies using clinical samples, and not reflecting the fact that there is also a high frequency of trauma in the general population (i.e. there are many individuals who mature into well-functioning adults, Paris, 1998). Therefore, this suggests that there is likely to be some protective factors serving these individuals. Resiliency is a key factor in protecting from the development of personality disorder. Werner and Smith (1992), for example, suggest that resiliency is related to adaptive personality traits in the individual. These traits enable them to form secure bonds. Factors associated with resiliency may include intellectual ability, ÷easyø temperament (i.e. low anxious individuals), autonomy, and communication skills (e.g., Baldwin et al., 1993; Brooks, 1994; Werner, 1982). The social background, which includes the family of the individual (e.g., Brooks, 1994; Cowen & Work, 1988) and the environment such as positive school experiences and peer relationships (e.g., Rutter, 1987; Wright &

Masten, 1997) may also increase resiliency Similarly, Rutter and Rutter (1993) propose that individuals who do not develop personality disorders are more likely to have had positive life events, which serve to safeguard against the trauma.

Finally, as dynamic risk factors consider risk management the issue of treatment adherence and its link to risk factors needs to be considered. As such there is a question whether risk assessments have secondary function in predicting treatment adherence. There is an ongoing need for further research into dynamic risk factors including personality disorder and key developmental variables and their relationship to treatment adherence.

CHAPTER 2

Investigating The Relationship Between Key Developmental

Variables, Personality Disorder, and Dynamic Risk assessment in

Sexual Offenders.

ABSTRACT

The area of risk assessment has grown considerably, particularly in relation to which factors best predict recidivism (Hanson & Morton-Bourgon, 2004), where dynamic factors have now been highlighted. Similarly, personality disorder has been identified as a risk factor for recidivism. More recently, key developmental variables have been associated with sex offender recidivism risk (e.g., Craissati, 2002). Craissati questions whether key developmental variables, personality disorder, and dynamic risk factors are, in fact, measuring a similar construct. In addition, there is a question whether dynamic risk assessments have secondary function in predicting treatment adherence. The aim of the current research is to extend Craissation work by identifying whether there is a relationship between personality disorder, as measured by the Millon Multiaxial Clinical Inventory (MCMI-III) (Millon, Millon, Davis, & Grossman, 1997), key, developmental variables, and dynamic risk domains, as measured by the Stable-2007 (formally Stable-2000 and Sex Offender Needs Assessment Rating) (Hanson & Harris, 2001; Hanson, Harris, Scott, & Helmus, 2007). Secondly, it aims to establish if these variables can be used as reliable markers of treatment attrition. 106 sex offenders were assessed, 69 child abusers and 37 rapists. Those participants who were assessed for treatment but who did not enter into the programme for any reason (i.e. those assessed only) were excluded from the treatment attrition statistical analyses. Chi-squared analyses were conducted to investigate relationships between the assessment modes and Area Under the Curve analysis was conducted to assess the predictive qualities of the variables on treatment attrition. In addition, due to the low sample size Power Analysis has been reported. Results indicated that there were a number of relationships between personality disorder, key developmental variables,

and items on the Stable-2007. However, very little was significant at predicting treatment attrition. The research had a number of methodological limitations largely due to the data collection methods and the limited sample size. The research concludes that this study demonstrates associations between key developmental variables, personality disorder and dynamic risk and that developmental variables should be considered in offender assessment and treatment. Further investigation in this area is warranted employing more rigorous research designs.

INTRODUCTION

The literature associated with the topics investigated in this empirical research have been reviewed in detail in Chapter 1 of this thesis. Therefore, in order to prevent repetitiveness but to provide a contextual background to the research what will be presented here is a brief introduction to the research or summary of the literature presented in Chapter 1.

Risk assessment of sexual offenders has been a pertinent topic for research and clinical developments for a number of decades. The serious societal impact and the impact on the victims has been a drive for much of the research into accurate risk assessment tools. Actuarial risk assessment tools (e.g. Static-99) (Hanson & Thornton, 2000), (Risk Matrix-2000) (Thornton et al., 2003) have been seen as the best available measures, receiving predictive accuracies in the moderate to good ranges (Hanson, 2004). However, actuarial risk assessments are not without their criticisms. One important critique of actuarial risk assessments is that they do not consider variables amenable to change and therefore cannot make suggestions for the treatment and management of sexual offenders. In response to this, a number of authors have investigated dynamic risk factors, that is, those amenable to change. Hanson and Harris (2001) further divided these into stable dynamic risk variables, which are persistent over months or years and acute dynamic risk variables, which can change in a matter of hours or even minutes. Furthermore, a number of dynamic risk assessment tools have been developed, which are demonstrating moderate predictive accuracy (e.g., Hanson, et al., 2007). Hanson et al. (2007) advocate for a comprehensive risk assessment tool, which comprises static, stable dynamic, and acute dynamic risk variables. However, more research into the variables included in such a risk assessment and the predictive accuracies of this tool is required.

Sexual offender personality and psychopathy have both been considered as risk variables in sexual offenders (e.g. Borchard, Gnoth, & Schlz, 2003; Chantry & Craig, 1994; Forth & Kroner, 1995). However, what has not been clear is whether a range of personality variables are associated with sexual offending as much of the literature has focussed on antisocial personality disorder (e.g. Hanson, 2000) and psychopathy (e.g., Hanson and Morton-Bourgon, 2004). There is a need for further research into a range of personality disorders and psychopathic variables in relation to sexual offence recidivism. In addition, there has been research into the role of key developmental variables and early attachment related to personality development and risk of sexual offending. Insecure attachments have been demonstrated to be evident in sexual offenders (Smallbone & Dadds, 1998; Ward, Hudson, & Marshall, 1996). Such attachments in childhood are associated with later personality development, such as the development of low self-esteem and difficulties forming intimacy (Marshall, Hudson, & Hodkinson, 1993), known risk factors of recidivism. Craissati and Beech (2006), using data from the Challenge Project, have proposed a model for assessing sex offending risk incorporating key developmental variables and static risk prediction. This research highlights the possibility that key developmental variables, personality, and dynamic risk variables may be related by a similar underlying construct. In addition, variables associated with risk of sexual offence recidivism, such as personality and psychopathy, have been associated with treatment failure (Craissati, Webb, & Keen, 2005). However, the empirical research linking risk

variables to treatment adherence variables remains sparse. This highlights a serious gap in the research and future research in this area is warranted.

The current research aims to identify the relationships, if any, personality disorder, as measured by the MCMI-III, key developmental variables, and dynamic risk domains, as measured by the Stable-2007. The research also aims to demonstrate whether these variables are related to treatment attrition in a group of community based sexual offenders (the Challenge Project).

Descriptions of key developmental variables can be found in Appendix 3 and details of the measures used in the current research can be found in Appendix 4.

Hypotheses

- 1. There will be significant associations between items on the Stable-2007, key developmental variables, and personality disorder on the MCMI-IIIø, as discussed in Chapter 1.
- 2. ÷The Stable-2007, key developmental variables, and personality disorder will be significantly associated with failure of the Challenge Programmeø
- 3. ÷The Stable-2007, key developmental variables, and personality disorder together will be predictive of failure of the Challenge Programmeø

METHOD

Sample

106 sex offenders both child molesters (n=69) and rapists (n=37) were referred for the Challenge project, a community based sex offender treatment programme. Participants were referred to the challenge project either at the time of sentencing or prior to release from custody for psychological reports. These offenders represent a specialist sample in that the offenders referred to the Challenge project are those offenders who present as challenging to manage in some way, whether this is by having a diagnosed mental illness, personality disorder, or previous non-compliance with supervision.

Procedure

An existing database was utilised which contained information on all participants gathered from probation files and entered onto a research schedule. Additionally, information was gathered through a semi-structured interview, covering areas of personal and social details, a full history of offending behaviour, and monitoring attitudes towards the offence. All background and offending variables were clearly defined in this process, but were also checked by the previous researchers, particularly if there were any doubts about the rating. Definitions of key developmental variables can be found in Appendix 3.

In addition, a number of standardised tests were administered. The Risk Matrix-2000 (Thornton et al., 2003), Static-99 (Hanson & Thornton, 2000), the Psychopathy Checklist-Revised (PCL-R) (Hare, 1991), and the Million Multiaxial Clinical

Inventory-Third Edition (MCMI-III) (Millon, Millon, Davis, & Grossman, 1997) were administered by previous authors and information pertaining to this was added to the database. The Stable-2007 was administered by the current primary author. This information was collected retrospectively through file information. Any information missing or unconfirmed in the files was obtained through asking key professionals who have knowledge of the client. Information on the Stable-2007 was conducted blind to ensure the author did not have information relating to success or failure of the treatment programme. Descriptions of each of the measures can be found in Appendix 4. Participants who were deemed suitable for treatment were either placed into group or individual (cognitive behavioural or supportive psychotherapy) community based treatment with appropriately qualified professionals (see Craissati & Beech, 2001 for details of the treatment programme). In order to analyse failure of the programme, participants who did not go on to receive treatment were disregarded from the sample at this point. This data was already available from previous authors and included formal failure only. Formal failure included; sexual, violent, or general re-convictions, and breach or recall. The data was then coded and appropriately analysed using the Statistical Package for the Social Sciences-Version 15 (SPSS-15). In order to test hypothesis one, Chi-squared analyses were employed. To test hypothesis two, Chisquared analysis was employed and a regression analysis was planned but was unsuccessful in execution due to the limited associations found in hypothesis one. To test hypothesis three, Area Under the Curve (ROC) analyses were used and due to small sample sizes Power Analysis was employed.

Ethical permission for the research was obtained by the London Probation Area by the Challenge Programme.

RESULTS

The sample comprised 106 sexual offenders, of which 69 (65.1%) were child sex offenders and 37 (34.9%) were adult sex offenders (rapists). Background and offending characteristics of the whole sample are outlined in Table 1. It can be seen from Table 1 that background variables demonstrate that the adult offenders were significantly younger than the child offenders. Both the child and adult offenders have experienced relatively high levels of childhood disturbances. However, the sample demonstrates child sex offenders experienced significantly more sexual abuse in childhood and more child sex offenders attended special schooling. No significant differences were observed in adulthood or offence related variables.

 ${\bf Table~1:~Background~and~offence-related~variables~for~the~sample}$

Referral agent:	Variable	All (%)	CM (%)	Rapists (%)
Referral agent:	Background variables	N=106	N=69 (65.1)	N=37 (34.9)
■ Solicitor 4 (3.8) 4 (5.8) 0 ■ Court 1 (0.9) 1 (1.4) 0 ■ Adult psychiatric service 5 (4.7) 3 (4.3) 2 (5.4) ■ Probation 87 (82.1) 53 (76.9) 34 (91.9) ■ Social services 2 (1.9) 2 (2.9) 0 ■ Mapps 2 (1.9) 2 (2.9) 0 ■ Other 4 (3.8) 3 (4.3) 1 (2.7) Childhood Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 17 (18.9)* Taken into local authority care	Age	41 sd 14	44 sd 13	35 sd 11**
■ Court 1 (0.9) 1 (1.4) 0 ■ Adult psychiatric service 5 (4.7) 3 (4.3) 2 (5.4) ■ Probation 87 (82.1) 53 (76.9) 34 (91.9) ■ Social services 2 (1.9) 2 (2.9) 0 ■ Mapps 2 (1.9) 2 (2.9) 0 ■ Other 4 (3.8) 3 (4.3) 1 (2.7) Childhood Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services<	Referral agent:			
■ Adult psychiatric service 5 (4.7) 3 (4.3) 2 (5.4) ■ Probation 87 (82.1) 53 (76.9) 34 (91.9) ■ Social services 2 (1.9) 2 (2.9) 0 ■ Mapps 2 (1.9) 2 (2.9) 0 ■ Other 4 (3.8) 3 (4.3) 1 (2.7) Childhood Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)*** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood 7 (48.1) 7 (48.9)	Solicitor	4 (3.8)	4 (5.8)	0
■ Probation	Court	1 (0.9)	1 (1.4)	0
■ Social services ■ Mapps □ Other □ O	 Adult psychiatric service 	5 (4.7)	3 (4.3)	2 (5.4)
■ Mapps 2 (1.9) 2 (2.9) 0 Childhood 4 (3.8) 3 (4.3) 1 (2.7) Childhood Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7)* Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)*** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood 12 (11.3) 10 (14.5) 2 <t< td=""><td>Probation</td><td>87 (82.1)</td><td>53 (76.9)</td><td>34 (91.9)</td></t<>	Probation	87 (82.1)	53 (76.9)	34 (91.9)
■ Other 4 (3.8) 3 (4.3) 1 (2.7) Childhood Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood 70 (56.6) 70 (76.8) <td>Social services</td> <td></td> <td>2 (2.9)</td> <td>0</td>	Social services		2 (2.9)	0
Childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (Mapps	2 (1.9)	2 (2.9)	0
Emotional/physical neglect in childhood 60 (56.6) 41 (59.4) 19 (51.4) Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables 8 (7.5) 4 (5.8) 4 (10.8) Remanded	Other	4 (3.8)	3 (4.3)	1 (2.7)
Physical abuse in childhood 40 (37.7) 29 (42) 11 (29.7) Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables 8 (7.5) 4 (5.8) 4 (10.8) ■ Remanded 14 (13.2) 11 (15.9) 3 (8.1) ■ Informal 9 (8.5) 7 (10.1) 2 (5.4) ■ Parole	Childhood			
Sexual abuse in childhood 36 (34) 29 (42) 7 (18.9)** 2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: *		60 (56.6)	41 (59.4)	19 (51.4)
2+ childhood disturbances 54 (50.9) 37 (53.6) 17 (45.9) Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: * Sentenced 8 (7.5) 4 (5.8) 4 (10.8) * Remanded 14 (13.2) 11 (15.9) 3 (8.1) * Informal 9 (8.5) 7 (10.1) 2 (5.4) * Parole 64 (60.4) 38 (55.1) 26 (70.3) * Probation 7 (6.6) 6 (8.7) 1 (2.7) <td></td> <td>40 (37.7)</td> <td>29 (42)</td> <td>11 (29.7)</td>		40 (37.7)	29 (42)	11 (29.7)
Taken into local authority care 31 (29.2) 18 (26.1) 13 (35.1) Special schooling 12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables Legal status: Sentenced 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Sexual abuse in childhood	36 (34)	29 (42)	7 (18.9)**
12 (11.3) 11 (15.9) 1 (2.7)* 1 or more trauma 48 (73.6) 53 (76.8) 25 (67.6) 2 or more trauma 60 (56.6) 43 (62.3) 17 (45.9) Adulthood Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status:	2+ childhood disturbances	54 (50.9)	37 (53.6)	17 (45.9)
1 or more trauma	Taken into local authority care	31 (29.2)	18 (26.1)	13 (35.1)
2 or more trauma	Special schooling	12 (11.3)	11 (15.9)	1 (2.7)*
Adulthood 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) • Remanded 14 (13.2) 11 (15.9) 3 (8.1) • Informal 9 (8.5) 7 (10.1) 2 (5.4) • Parole 64 (60.4) 38 (55.1) 26 (70.3) • Probation 7 (6.6) 6 (8.7) 1 (2.7)	1 or more trauma	48 (73.6)	53 (76.8)	25 (67.6)
Contact with mental health services 12 (11.3) 10 (14.5) 2 (5.4) History of self harm 9 (8.5) 6 (8.7) 3 (8.1) No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) • Remanded 14 (13.2) 11 (15.9) 3 (8.1) • Informal 9 (8.5) 7 (10.1) 2 (5.4) • Parole 64 (60.4) 38 (55.1) 26 (70.3) • Probation 7 (6.6) 6 (8.7) 1 (2.7)	2 or more trauma	60 (56.6)	43 (62.3)	17 (45.9)
History of self harm	Adulthood			
No co-habiting relationships 60 (56.6) 42 (60.9) 18 (48.6) Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Contact with mental health services	12 (11.3)	10 (14.5)	` ′
Never married 51 (48.1) 31 (44.9) 21 (56.8) Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	History of self harm	9 (8.5)	6 (8.7)	3 (8.1)
Employment problems 74 (69.8) 46 (66.7) 28 (75.7) Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)		` /	42 (60.9)	18 (48.6)
Offence-related variables All (%) CM (%) Rapists (%) Legal status: 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Never married		31 (44.9)	21 (56.8)
Legal status: Sentenced 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Employment problems	74 (69.8)	46 (66.7)	28 (75.7)
Sentenced 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Offence-related variables	All (%)	CM (%)	Rapists (%)
Sentenced 8 (7.5) 4 (5.8) 4 (10.8) Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Legal status:			
Remanded 14 (13.2) 11 (15.9) 3 (8.1) Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	-	8 (7.5)	4 (5.8)	4 (10.8)
Informal 9 (8.5) 7 (10.1) 2 (5.4) Parole 64 (60.4) 38 (55.1) 26 (70.3) Probation 7 (6.6) 6 (8.7) 1 (2.7)	Remanded	` ′	, ,	' '
Parole 64 (60.4) 38 (55.1) 26 (70.3) 7 (6.6) 6 (8.7) 1 (2.7)	Informal	9 (8.5)	, ,	' '
■ Probation 7 (6.6) 6 (8.7) 1 (2.7)		· ·	` ′	' '
	Probation	, ,	, ,	' '
	Previous sexual convictions	` ′	, ,	, ,
Unconvicted allegations (sexual) 27 (25.5) 19 (27.5) 8 (21.6)		` ′	` ′	` ′
Victim gender (any sexual conviction):		, ,		
■ Male 21 (19.8) 19 (27.5) 2 (5.4)		21 (19.8)	19 (27.5)	2 (5.4)
Female 70 (66) 37 (53.6) 33 (89.2)		· · ·	` ,	' '
■ Both 10 (9.4) 10 (14.5) 0		, ,	` ,	` ′
Substance abuse at time of offence: 49 (46.2) 26 (37.7) 23 (62.6)			` ,	23 (62.6)

^{*} p<.05, ** p<.01

Risk

The results of the psychometric measures for risk are detailed in Table 2. It can be seen from Table 2 that no significant differences were observed between child sex offenders and adult offenders on the Risk Matrix-2000 or the Static-99. There were very few child sex offenders or adult offenders who were defined as psychopaths on the PCL-R (score of above 25). Scores for the risk domains on the Stable-2007 were rated as present if a score of definitely present or possibly present was given. There was a prevalence of the sample scoring as present in the significant social influences, relationship instability, poor problem solving skills, and deviant sexual preferences risk domains. However, even when a more rigorous cut off was used (0, 1 = not present; 2 = present), there was still a high prevalence of difficulties observed within these variables. On the Stable-2007 variables significant differences were observed within significant social influences and deviant sexual preferences, with child molesters being more likely to be scored as present on these factors. There were no significant differences between child and adult sexual offenders on the risk categories of the Stable-2007. There were also no significant differences on the total score of the Stable-2007.

Table 2: Results of the psychometric measures for risk

Psychometric measures (risk)	All (%) N=106	CM (%) N=69	Rapists (%) N=37
Risk Matrix 2000:			
■ Low	18 (17)	12 (17.4)	6 (16.2)
Medium	35 (33)	25 (36.2)	10 (27)
■ High	27 (25.5)	14 (20.3)	13 (35.1)
Very high	18 (17)	11 (15.9)	7 (18.9)
Static 99:	, ,	, , ,	
■ Low	16 (15.1)	14 (20.3)	2 (5.4)
 Medium low 	28 (26.4)	17 (24.6)	11 (29.7)
Medium high	30 (28.3)	16 (23.2)	14 (37.8)
■ High	25 (23.6)	16 (23.2)	9 (24.3)
PCL-R:	(
Present	3 (2.8)	39 (56.5)	20 (54.1)
Not present	59 (55.7)	2 (2.9)	1 (2.7)
Stable-2007 ¹ (n=78):		- ()	,
 Significant influences definite 	18 (17)	14 (20.3)	4 (10.8)
 Significant influences some 	54 (50.9)	40 (58)	14 (27)*
 Relationship instability definite 	42 (39.6)	31 (44.9)	11 (29.7)
 Relationship instability some 	64 (60.4)	44 (63.8)	20 (54.1)
 Emotional ID definite 	15 (14.1)	14 (20.3)	N/A
 Emotional ID some 	32 (30.2)	31 (44.9)	N/A
 Hostility to women definite 	9 (8.5)	5 (7.2)	4 (10.8)
 Hostility to women some 	25 (23.6)	14 (20.3)	11 (29.7)
 General social rejection definite 	24 (22.6)	19 (27.5)	5 (13.5)
 General social rejection definite General social rejection some 	51 (48.1)	37 (53.6)	14 (37.8)
 Lack of concern for others definite 	17 (16)	13 (18.8)	4 (10.8)
 Lack of concern for others some 	35 (33)	24 (34.8)	11 (29.7)
	16 (15.1)	13 (18.8)	3 (8.1)
impulsivity definite	` /	, ,	, ,
impulsivity some	` /	` /	` /
 Poor problem solving definite Poor problem solving definite 	` /	15 (21.7) 39 (56.5)	` /
1 oor problem sorving definite	57 (53.8)	` ′	` /
 Negative emotionality some 	18 (17)	13 (18.8)	5 (13.5)
Negative emotionality some	38 (35.8)	27 (39.1)	11 (29.7)
• Sexual preoccupations definite	13 (12.3)	8 (11.6)	5 (13.5)
• Sexual preoccupations some	29 (27.4)	21 (30.4)	8 (21.6)
 Sex as coping definite 	16 (15.1)	10 (14.5)	6 (16.2)
• Sex as coping some	34 (32.1)	20 (29)	14 (37.8)
 Deviant sexual preferences definite 	26 (24.5)	19 (27.5)	7 (18.9)
 Deviant sexual preferences some 	55 (51.9)	43 (62.3)	12 (32.4)**
 Co-operation with supervision definite 	6 (5.7)	4 (5.8)	2 (5.4)
 Co-operation with supervision some 	31 (29.2)	20 (29)	11 (29.7)
■ Low	8 (7.5)	5 (7.2)	3 (8.1)
 Moderate 	36 (34)	22 (31.9)	14 (37.8)
	30 (28.3)	23 (33.3)	7 (18.9)
HighTotal Score	11 sd 5	11 sd 5	9 sd 6

^{*}p<.05, ** p<.01, 1 Definite/some

Personality

The results of the MCMI-III are laid out in Table 3 (personality traits).

When viewing the frequencies and percentages of the presence of personality disorder in child abusers and rapists a few differences can be observed. However, statistical analysis revealed no significant differences between the two groups. A moderate level of personality difficulties were reported overall with 37% of the sample reporting dysfunction and 21% reporting dysfunction to a level considered to be diagnosable personality disorder.

Table 3: Results of the MCMI-III

Variable	All N=66 (%)	CM N=49 (%)	<i>Rapists</i> N=17 (%)
Schizoid PD	9 (8.5)	7 (10.1)	2 (5.4)
	26 (24.5)	21 (30.4)	5 (13.5)
Definite/probable ¹			
Paranoid PD <i>Definite</i> Paranoid PD <i>probable</i>	3 (2.8) 9 (8.5)	3 (4.3) 9 (13)	0 0
Schizotypal PD <i>Definite</i>	3 (2.8)	3 (4.3)	0 1 (2.7)
Schizotypal PD <i>probable</i>	9 (8.5)	8 (11.6)	
Antisocial PD <i>Definite</i> Antisocial PD <i>probable</i>	3 (2.8)	2 (2.9)	1 (2.7)
	12 (11.3)	8 (11.6)	4 (10.8)
Borderline PD <i>Definite</i> Borderline PD <i>probable</i>	3 (2.8)	2 (2.9)	1 (2.7)
	11 (10.4)	7 (10.1)	4 (10.8)
Histrionic PD <i>Definite</i> Histrionic PD <i>probable</i>	5 (4.7)	4 (5.8)	1 (2.7)
	6 (5.7)	5 (7.2)	1 (2.7)
Narcissistic PD <i>Definite</i>	2 (1.9)	2 (2.9)	0 1 (2.7)
Narcissistic PD /probable	4 (3.8)	3 (4.3)	
Avoidant PD <i>Definite</i> Avoidant PD <i>probable</i>	5 (4.7)	5 (7.2)	0
	29 (27.4)	24 (34.8)	5 (13.5)
Dependent PD <i>Definite</i> Dependent PD/ <i>probable</i>	9 (8.5)	7 (10.1)	2 (5.4)
	29 (27.4)	23 (33.3)	6 (16.2)
Obsessive-compulsive <i>Definite</i> Obsessive-compulsive <i>probable</i>	3 (2.8)	1 (1.4)	2 (5.4)
	5 (4.7)	2 (2.9)	3 (8.1)
Sadistic <i>Definite</i> Sadistic <i>probable</i>	8 (7.5) 8 (7.5)	8 (11.6) 8 (11.6)	0 (0) 0 (0)
Negativistic <i>Definite</i> Negativistic <i>probable</i>	14 (13.2)	12 (17.4)	2 (5.4)
	25 (23.6)	18 (26.1)	7 (18.9)
Self-Defeating <i>Definite</i> Self-Defeating <i>probable</i>	12 (11.3)	11 (15.9)	1 (2.7)
	26 (24.5)	20 (29)	6 (16.2)
Depressive <i>Definite</i> Depressive <i>probable</i>	6 (5.7) 9 (8.5)	6 (8.7) 9 (13)	0
Cluster A <i>Definite</i>	11 (10.4)	9 (13)	1 (2.7)
Cluster A <i>probable</i>	35 (33)	24 (34.8)	6 (16.2)
Cluster B <i>Definite</i>	12 (11.3)	10 (14.5)	2 (5.4)
Cluster B <i>probable</i>	27 (25.5)	23 (33.3)	4 (10.8)
Cluster C <i>Definite</i>	8 (7.5)	7 (10.1)	1 (2.7)
Cluster C <i>probable</i>	25 (23.6)	19 (27.5)	6 (16.2)

^{*} p<.05, ** p<.01
Definite=score of 85+, Probable=score of 75+

Developmental Variables, Personality Disorder and Risk

Chi-squared tests were performed to test whether there were significant associations between developmental variables, personality disorder, and risk. However, a numbers of the cells had an expected cell count of less than five and as such Fishers Exact tests have been reported in these cases.

Significant associations between PD and Stable-2007

(The personality disorder clusters can be found in Appendix 6)

- General Social Rejection was associated with definite presence of Cluster B personality disorder (Fishers Exact p<.05), probable presence of Cluster C personality disorder (X² 4.4, df 1, p<.05), and probable presence of Paranoid Personality disorder (Fishers Exact p<.05).
- Sex as Coping was significantly associated with Histronic Personality Disorder (Fishers Exact p<.05), definite presence of Cluster B Personality Disorder (Fishers Exact p<.01), definite presence of Avoidant Personality Disorder (Fisher Exact p<.05), probable presence of Antisocial Personality Disorder (Fishers Exact p<.05), definite presence of Cluster B (Fisher Exact p<.05), and probable presence of Cluster C (Fisher Exact = p<.01).
- Relationship Instability was significantly associated with definite presence of Compulsive Personality Disorder (Fishers Exact p<.05) and definite presence of Masochistic Personality Disorder (Fisher Exact p<.05).

- Lack of Concern for Others was significantly associated with probable presence of Dependent Personality Disorder (X² 4.8, df 1, p<.05).
- Impulsivity was significantly associated with probable presence of Cluster C
 Personality Disorder (Fisher Exact p<.05).
- Deviant Sexual Preferences was significantly associated with definite presence of Schizoid Personality Disorder (Fishers Exact p<.05) and definite presence of Personality Disorder (Fishers Exact p<.01).
- Emotional Identification with Children was significantly associated with definite presence of Schizoid Personality Disorder (Fishers Exact = p<.05).

Significant associations between key developmental variables and Stable-2007

Due to the number of associations between these variables, the definite presence of items on the Stable-2007 has been reported.

- Significant Social Influences was significantly associated with emotionally/physically neglected (X^2 4.0, df 1, p<.05), sexual abuse (X^2 4.6, df 1, p<.05), and the presence of two or more childhood traumas (X^2 4.0, df 1, p<.05).
- Relationship Instability was significantly associated with emotionally/physically neglected (X² 7.1, df 1, p<.01).

- Emotional Identification with Children was significantly associated with sexual abuse (X^2 5.3, df 1, p<.05).
- General Social Rejection was significantly associated with emotionally/physically neglected (X² 4.2, df 1, p<.05).
- Lack of Concern for Others was significantly associated with physical abuse (X²
 9.7, df 1, p<.05), two or more abuses (X² 6.3, df 1, p<.05), and two or more childhood traumas (X² 6.7, df 1, p<.01).
- Impulsivity was significantly associated with one or more childhood traumas (Fishers Exact p<.05), and two or more childhood traumas (X^2 5.9, df 1, p<.05).
- Poor Problem Solving was significantly associated with emotionally/physically neglected (X^2 4.7, df 1, p<.05), physical abused (X^2 6.9, df 1, p= p<.01), one or more childhood traumas (Fishers Exact p<.05), and two or more childhood traumas (X^2 4.7, df 1, p<.05).
- Negative Emotionality was significantly associated with physical abuse (X^2 6.3, df 1, p<.05), sexual abuse (X^2 4.1, df 1, p<.05), and two or more childhood traumas (X^2 4.1, df 1, p<.05).
- Sex as Coping was significantly associated with one or more childhood traumas (Fishers Exact p<.05).

 Deviant Sexual Preferences was significantly associated with sexual abuse (X² 4.4, df 1, p<.05).

Significant associations between PD and key developmental variables

- Schizoid Personality Disorder was significantly associated with sexual abuse (Fishers Exact p<.05).
- Possible presence of Depressive Personality Disorder was significantly associated with physical abuse (Fishers Exact p<.01), two or more abuses (Fisher Exact p<.01) and definite presence of Depressive Personality Disorder was significantly associated with two or more abuses (Fishers Exact p<.05).
- Possible presence of Antisocial Personality Disorder was significantly associated with two or more abuses (Fishers Exact p<.05).
- Possible presence of Masochistic Personality Disorder was significantly associated with sexually abuse (X^2 4.6, df 1, p<.05), at least one trauma (X^2 5.0, df 1, p<.05).
- Cluster B personality disorders was significantly associated with sexually abuse
 (X² 7.1, df 1, p<.01) and physically abused (X² 6.4, df 1, p<.05).
- Cluster C personality disorders was significantly associated with two or more abuses (X^2 4.4, df 1, p<.05),

Treatment Attrition

Of the total sample, 31 failed to complete the programme, 21 child sex offenders failed and 10 rapists. There were no significant differences between the child sex offenders and the rapists on failure/survival. Chi squared analyses were performed to explore any associations between key developmental variables, personality disorder, and/or items on the Stable-2007 and failure of the Challenge Programme. A regression analysis was then planned to build a model predictive of treatment failure. However, this could not be achieved due to the limited associations between the variables.

AUC Analysis

AUC analysis was performed to test for any significant predictive value of personality disorder, developmental variables, and Stable-2007 scores to failure of the Challenge programme. Only one significant predictive value was observed on the probable presence of Cluster A personality disorder (AUC = 0.84, CI = .69 - .99, p>0.05) and two moderate predictive values were observed on the Avoidant personality disorder (AUC = 0.75, CI = .44 - .89, non-significant) and Schizotypal personality disorder variable (AUC = 0.75, CL = .40 \(\text{o} \) 1.0, non-significant). Although it can be seen that this was non-significant (p>0.05), when the 95% confidence intervals are viewed the results indicate that they would be more promising if there was a larger sample. In addition, based on Craissati, Webb, and Keen (2005) findings, when new variables are computed to attempt to increase the predictive value of the Stable-2007, personality disorder, and key developmental variables, no significant predictive values are observed.

Table 4 ROC results:

			Asymptotic	• •	5% Confidence
Test Result Variable(s)	Area	Std. Error ^a	Sig.b	Lower Bound	Upper Bound
Millon schizoid	.67	.12	.22	.44	.89
Millon avoidant	.75	.10	.07	.54	.95
Millon depressive	.53	.14	.86	.25	.80
Millon dependent	.68	.14	.18	.41	.96
Millon histrionic	.60	.12	.48	.37	.82
Millon narcissistic	.61	.12	.41	.37	.86
Millon antisocial	.56	.16	.67	.25	.86
Millon schizotypal	.75	.13	.06	.49	1.0
Millon aggressive	.60	.15	.48	.31	.88
Millon borderline	.66	.14	.25	.39	.93
Millon paranoid	.66	.13	.25	.41	.90
Millon compulsive	.64	.14	.30	.36	.92
Millon passive aggressive	.59	.14	.52	.31	.86
Millon self defeating	.55	.13	.74	.29	.81
Cluster A probable	.84	.08	.02	.69	.99
Cluster A definite	.61	.13	.48	.35	.86
Cluster B probable	.59	.15	.55	.31	.87
Cluster B definite	.48	.15	.89	.19	.77
Cluster C probable	.33	.14	.26	.05	.61
Cluster C definite	.50	.15	1.0	.21	.79
Probable PD	.46	.15	.80	.18	.75
Definite PD	.66	.12	.29	.42	.89
Emotionally/physically neglected	.51	.11	.65	.31	.80
Physically abused	.49	.11	.89	.27	.70
Sexually abused	.54	.10	.68	.34	.74
Two or more abuses	.52	.11	.87	.30	.74
At least 1 trauma	.47	.10	.75	.27	.67
At least 2 trauma	.56	.11	.57	.35	.77
Total score on Stable- 2007	.57	.13	.65	.31	.80
Stable 2007+PD	.40	.16	.57	.09	.72
Stable 2007+Dev. vars.	.55	.13	.72	.28	.81
PD+Dev. Vars.	.51	.15	.92	.23	.81

Power Analysis

In light of the fact that there were limited predictive values of the Stable-2007, personality disorder, and key developmental variables (which is contrary to previous findings) and that the current sample size was relatively small, power analysis was performed in order to determine whether predictive values could be increased given a larger sample size. This was completed using the G Power statistical package. Power analysis was performed on those variables which achieved an AUC of more than 0.5. Results indicated that many of the variables would need a sample size of over 1000 to achieve a significant effect. Other variables appeared more promising, requiring sample size of around one or two hundred, for example the definite presence of personality disorder required a sample size of 79, paranoid personality disorder 101, definite presence of Cluster A personality disorder 131, and sadistic and probable presence of Cluster B personality disorder requiring a sample size of 178.

DISCUSSION

This empirical research aimed to investigate the relationships between dynamic risk variables, personality disorder, and key developmental variables. In addition, it aimed to demonstrate whether any of these variables were related to treatment attrition in a sample of community based sex offenders. The sample included 106 sexual offenders (69 child sex offenders and 37 rapists). There were relatively few differences in background and offence related variables within the sample. There were also relatively few differences on psychometric measures for risk with no significant differences of child sexual offenders and rapists being observed on risk level of the measures. The sample revealed moderate levels of personality difficulties. However, no significant differences were observed between the groups. In the study there was a potential for Type 1 errors, i.e. a null hypothesis being incorrectly rejected. This could be reduced by increasing from a 0.05 to 0.01 significance level.

It was hypothesised that 1. :There will be significant associations between items on the Stable-2007, key developmental variables, and personality disorder on the MCMI-III, 2. :The Stable-2007, key developmental variables, and personality disorder will be significantly associated with failure of the Challenge Programmeø and 3. :The Stable-2007, key developmental variables, and personality disorder will be predictive of failure of the Challenge Programmeø

Key findings

The current research findings provided a number of results supportive of Hypothesis One. Although total score on the Stable-2007 was not associated with any of the personality disorder variables, Chi-squared and Fishers Exact tests did demonstrate that many of the items on the Stable-2007 were associated with some of the personality disorders and the personality disorder clusters. In addition, the Stable-2007 items and personality disorder variables were significantly associated with key developmental variables. In terms of Hypothesis Two, this was not supported by the research findings as Chi-squared analyses revealed that there were no associations between Stable-2007, key developmental variables, and personality disorder and failure of the Challenge Programme. However, this may have been affected but the low sample size of offenders who failed to complete the programme. Hypothesis Three was partially supported by the research findings. AUC analysis demonstrated was one significant predictive value was observed on the probable presence of Cluster A personality disorder. However, more promising results were expected for a number for the variables (definite presence of PD, paranoid PD, definite presence of Cluster A PD, sadistic PD, and probable presence of Cluster B PD) if a larger sample size were to be used, as demonstrated by G Power, power analysis.

These findings would have been expected and are in line with previous research. Sexual offenders have been consistently found to experience trauma and neglect in childhood (Bard et al., 1987; Craissati & McClurg, 1996), which is evident in the current findings. Furthermore, sexual abuse has been previously found to be associated with later sexual offending (Craissati, et al., 1998). Other developmental variables previous associated with sexual offending physical, sexual, or emotional

abuse, higher levels of childhood difficulties such as being bullied and having difficulties in friendships, resulting in adult characteristics such as self harm, hostility, lack of empathy, sexual pre-occupation, and contact with mental health services (Craissati, McClurg, & Browne, 2002a). In terms of treatment compliance, the notion that the current findings may be significant if there was an increased sample size is supported by previous findings. For example, Craissati and Beech (2001) indicate that treatment attrition is associated with childhood difficulties and childhood victimisation. In terms of the predictive value of the variables this has also been supported by previous research. While, Harris, Rice, and Cormier (2002) indicated that personality disorder was related to sexual offence recidivism. Moreover, a risk prediction model, which includes static or dynamic risk, the presence of childhood trauma, and mental health difficulties or personality disorder, has been found to be predictive of sexual offence recidivism (Craissati, Webb, & Keen, 2005).

Limitations and suggestions for future research

The current research is not without its limitations. Firstly, as reported in the results the limited sample size affected the number of analyses that could be performed as well as the quality of these analyses. Future research using a larger sample size would certainly be useful and would be feasible given that the data continues to be collected at the Challenge Project. Furthermore, research into Personality disorder and key developmental variables potentially measuring the same construct would be useful. This could be achieved by partially replicating the current research, whilst developing it to alleviate the methodological limitations. The research design included a number of methodological limitations and these may be alleviated by considering the following: The data collection was affected by some missing information on the

research schedules. This was partially alleviated by contacting report authors. However, this would have obvious limitations with historical collection of data as the report authors would often be recalling from memory rather than documented information. This may be alleviated in future by using a longitudinal design rather than retrospective for this data collection. The data collection and reports were completed by different professionals, which has problems associated with inter-rater reliability. In particular, it was obvious that some professionals provided clearer information on dynamic variables within the psychological reports than others did. This was an interesting observation as it seemed that the more recent psychological reports included more detailed and structured information on dynamic risk variables. It may be that this is a reflection of the current research and clinical climate, where dynamic variables are becoming viewed as more pertinent in the overall risk assessment of sexual offenders. This may be overcome by ensuring all report authors have the same training level and are up to date with current research. Further, background information of the offenders on the research schedules was collected via self-report. The MCMI-III is also a self-report measure for personality functioning. Similarly, the Stable-2007 was collect by file information, which at the point this information was initially obtained would have been largely self-report. It may be that the offenders self-report includes a minimisation or exaggeration of the difficulties experienced. This may be at least partially alleviated in future research if a DSM-IV diagnosis was also used to determine personality disorder. However, much research and clinical information is gathered in this way so there is limited evidence to suggest that the current research is anymore flawed in this sense than is standard.

The study would also have benefited from using a control group, for example from the local area probation service. However, the data for a control group was not available for the current research. Further, non-contact offenders were excluded from the current research, although they are accepted for treatment on the Challenge Programme. This was due to the low number of non-contact offenders to comprise heterogeneous group and the researchers did not want to categorise them as rapist or child sexual offenders. As the data collection on the Challenge Project has continued it would be useful to consider this group of offenders in future research projects. Finally, due to the nature of the research sample, i.e. those sexual offenders who have been referred to a specialist treatment programme, the applicability of the research findings to other settings will inevitably be somewhat limited. Future research might consider a non-specialist treatment group for further comparisons. This may include, as suggested previously, local probation samples. It may also be valuable to consider inpatient and prison samples for comparison. Furthermore, it may be of interest to include in a future research design general and violent offenders. The Stable-2007 would not be useful in this as it is a dynamic risk assessment for sexual offenders only. However, other dynamic risk assessment could be used and evaluation of whether the associations between risk level personality disorder and key developmental variables is specific to sexual offenders or the current cohort could occur.

Conclusions

The research presented is preliminary piece of work and clearly more work needs to be completed in this area, following above recommendations. The research does demonstrate that further investigation is warranted and it gives weight to theory that key developmental variables, personality disorder, and dynamic risk assessments may be measuring similar underlying constructs. In addition, it highlights the importance for professionals to consider assessment and treatment of sexual offenders from a developmental perspective. More rigorous research designs with larger samples sizes and comparison groups would improve upon the current research.

Chapter 3:

Case Study of a 41 Year Old High Risk Sexual Offender

CHAPTER 4

Critique of a Psychometric Measure: Stable-2000 (Hanson & Harris, 2000)

INTRODUCTION

Over the past decade, research pertaining to the area of sex offender risk assessment and recidivism has evolved considerably. Most notably, the literature has expanded in relation to which variables are the best predictors of future risk and offending behaviour (Hanson & Morton-Bourgon, 2004). This research base has identified that key predictors of sexual offence recidivism are static, unchangeable variables such as; age, sexual offence history, length of intimate relationships or cohabitation, stranger victims, and victim gender. Correspondingly, this has lead to the development of a number of actuarial risk assessment tools, designed to predict sexual offence recidivism, based upon static variables (Craig, Browne, Stringer & Beech, 2005). The most widely used of which are the Risk Matrix-2000 (Thornton et al, 2003) and the Static-99 (Hanson & Thornton, 2000).

Actuarial risk assessments are now well established in both clinical and research settings. Furthermore, to date, they have consistently demonstrated their efficacy in risk prediction, in particular by repeatedly providing at least moderate predictive power for sexual offence recidivism (AUC > .7) (Hanson, 2004). However, although these assessments are now well rooted within the empirical literature (e.g. Harris, Rice, & Cormier, 2002; Hood, Shute, Feilzer, & Wilcox, 2002) they fail to consider the role of risk management (Prins, 2005; Beech, Fisher, & Thornton, 2003). Furthermore, they do not guide clinicians on the psychological factors associated with risk and provide no indications of how risk can be reduced (Beech & Ward, 2003).

In order to alleviate some of the problems associated with static risk assessments, a number of researchers have focussed their attentions on dynamic risk factors, i.e. those factors which are amenable to change (Hanson & Harris, 2001). Theoretical and practical attempts have been made to introduce these dynamic variables and develop standardised assessment procedures. Such dynamic approaches maintain the importance of offence-related factors, whilst accounting for a broad range of psychological difficulties (Craissati, Webb, & Keen, 2005). The initial empirical base pertaining to dynamic factors suggests that there are four significant dynamic domains predictive of sexual offence recidivism; pro-offending attitudes, intimacy deficits, sexual self-regulation (including deviant sexual interests) and general self-regulation (e.g. Barbaree & Marshall, 1989; Hanson & Bussière, 1998; Beech, Fisher, & Beckett, 1999; Proulx, Perrault, & Ouimet, 1999.

Two notable dynamic risk assessment tools have currently been validated for use with sexual offenders, the Sex Offender Treatment Evaluation Project (STEP) test battery (Beech, Fisher, & Beckett, 1999) and the Initial Deviance Assessment (IDA) (Thornton, 2002). The STEP test battery, implemented within the U.K. probation service, determines pro-offending attitudes, socio-affective problems, and deviant sexual interest, through a number of psychometric measures and the IDA, used in U.K. Prison Service, focuses on: deviant sexual interest, distorted attitudes, socio-affective functioning, and problems with self-management. In both assessment measures the dynamic domains have been shown to be predictive of sexual offence recidivism, independently of static variables (Beech, Fisher & Thornton, 2003). Furthermore, preliminary research suggests that dynamic variables can enhance the accuracy of actuarial risk assessments (Hanson & Harris, 2001; Beech, Friendship,

Erikson & Hanson 2002; Thornton, 2002). Nonetheless, there is an ubiquitous need for continuing research into sexual recidivism and the development of comprehensive risk assessment tools, which comprise well-validated static and dynamic variables.

In accordance with this developing literature base, Hanson and Harris (2000) have developed the Stable-2000. This review will examine the Stable-2000, in terms of its scientific properties and its applicability to sexual offender risk assessment.

Overview of the stable 2000

The Stable-2000 is a 16 item measure of stable dynamic risk factors designed for use with adult sexual offenders. These factors are organised into 6 subsections (see Stableó2000 Tally Sheet, appendix 5). 1. Significant Social Influences, which refers to individuals in the offenders life who are not paid to be there. 2. Intimacy Deficits, which is divided into five items, a) stability of the offenders current intimate relationship; b) emotional identification with children; c) hostility toward women; d) general social rejection/loneliness; and e) lack of concern for others. 3. Sexual Self-Regulation, comprising three items; a) high sex drive/sexual pre-occupations; b) use of sex as a coping strategy; and c) deviant sexual interests. 4. Attitudes Supportive of Sexual Assault, which assesses three items a) sexual entitlement; b) attitudes tolerant of rape; and c) attitudes tolerant of adult-child sex. 5. Co-operation with Supervision, which identifies the degree to which the offender is co-operating with their supervision. 6. General Self-Regulation, comprising three items a) impulsive acts; b) poor cognitive problem-solving skills; and c) negative emotionality/hostility.

Each item is assessed following a structured scoring guide using three-point rating scales, 0 (no problem), 1 (some concern/slight problem) and 2 (present/definite concern). When all the items have been scored, the highest score in each subsection is calculated as the section score. The subsection scores are then tallied to arrive at an overall Stable-2000 score ranging from 0 to 12. In terms of risk categories total scores of 0-4 are considered to be low, scores of 5-8, moderate, and scores of 9-12, high. These score ranges are not validated and were derived from the clinical judgement of Hanson and Harris (2000), the tooløs authors. The Stable-2000 can be administered at any time during the offendersø supervision period. Therefore, it allows assessment of changes in risk/need to be made as the offender progresses through treatment or if significant, potentially risk altering, events occur.

The Stable-2000 does not have a professional manual. However, a comprehensive administration and scoring guide is available by emailing the authors. When initially developed the Stable-2000 was empirically informed but not validated. Since this date, attempts have been made to validate the measure (e.g. Hanson, Harris, Scott, & Helmus, 2007). However; a formal manual is still unavailable. As such, normative data is not comprehensively presented and information pertaining to factor structure and test construction is absent. Furthermore, as a result, although used in forensic practical settings, the literature pertaining to the measure is somewhat sparse. The Stable 2000 has since been developed into the Stable 2007. The differences between the measures lie in their validation, as the Stable-2007 has been further validated and that the items which were not predictive of sexual offence recidivism in the Stable 2000 have been removed from the Stable 2007. Furthermore, the Stable 2007 does not divide the individual items into subsections, rather it has one overall score obtained by

tallying all the items. The Stable 2000 was used as a critique in this measure due to the Stable 2007 being unpublished at the time of writing. In addition, it was felt useful to highlight these differences and the tools development.

RELIABILITY

Internal Consistency

A test is considered reliable if it is self-consistent (internal consistency). This is important as internal consistency relates to the degree to which the items are correlated to the criterion and how well each item relates to each other independently (Janda, 1994), usually measured by Cronbachøs alpha co-efficient. The internal consistency alpha for the Stable-2000 equalled .83. Field (2000) suggests that alphas over 0.60 reflect a measure that it internally consistent. Nunnally (1978) takes a more conservative view of alphas over 0.70 being an acceptable level. Which ever level is considered, the alpha provided demonstrates that the Stable-2000 has very good internal consistency. However, Hanson, Harris, Scott, & Helmus (2007) is the only paper which has reported on the internal consistency of the measure. It would enhance the reliability if additional studies were to report on the internal consistency. Moreover, the inter-item correlations are not reported on and as Cattell and Kline (1977) point out this could be anti-theoretical. In the absence of inter-item correlations it remains unknown whether each item on the Stable-2000 are operating in the same way to the criterion. In other words, if the inter-item correlations are too great then each item would not be providing any new information. Therefore, the measure would only pertain to a limited theoretical criterion with little variance.

Test-retest reliability

A test is also deemed reliable if it achieves adequate test re-test reliability. Test-retest reliability refers to correlating the scores from set of subjects tested on two separate occasions (Kline, 1986), where no intervention has been provided. Using correlational

analysis a minimum level of 0.7 should be reported. A correlation of less than this is deemed unsatisfactory as the standard error would become too large resulting in uncertain interpretations (Guilford, 1956). Unfortunately, no information pertaining to test-re-test reliability is available on the Stable-2000.

Inter-rater reliability

More recently, Hanson, Harris, Scott and Helmus (2007) have reported on the interrater reliability of the Stable-2000. The authors tested for inter-rater reliability in two ways. Firstly, they compared the responses 258 trained officers to their own responses. The authorsø responses were taken to be the correct answers. They demonstrated that for the Stable-2000 total scores, between 47% and 67% were within one point of the correct answer and between 74% and 99% were within three points of the correct answer. On initial examination, these ÷percent correctø scores appear to demonstrate a high degree of inter-rater reliability. However, on closer examination and when considered with the scoring guide a difference of two or three points on the total Stable score can affect an offenders risk category relatively easily. Therefore, it may be asserted that only the percent scores within one point of the correct answer can be used as a valid measure of inter-rater reliability. In doing this, it is evident that the inter rater reliability is reduced considerably, at times to less than 50% accurate.

Secondly, the authors file reviewed 92, randomly selected cases which had existing Stable-2000 scores. They identified the best scoring possible based upon the file information. The inter-rater reliability achieved here was high, the intra-class correlation was .89 (k = 87). However, the reliability may have been artificially affected as the second raters (the authors) were not blind to the original ratings and

they also took the opportunity to question the first raters about any missing or ambiguous information.

VALIDITY

Face Validity

Face validity refers to the extent to which a measure appears to be measuring what it claims to. Employing theoretical knowledge of the area, the Stable-2000 demonstrates good face validity as all the items appear to be relevant to sex offender risk of recidivism.

Predictive Validity

As with the reliability, predictive validity information is not available within the initial guide of the Stable-2000. However, the subsequent study (Hanson, Harris, Scott and Helmus, 2007) does provide some predictive validity information, using the criterion recidivism. They utilised Stable-2000 assessments of 805 adult male sex offenders and followed them up to a survival end date which was set one month after the last assessment information was received. They then recorded the predictive validity of five types of recidivism; sexual crime recidivismø (AUC = .64), sany sexual recidivismø (AUC = .66), sviolent recidivismø (AUC = .65), sany criminal recidivismø (AUC = .64), and sany recidivismø (AUC = .67). These results demonstrate that the Stable-2000 total scores provide good predictive validity on all five types of recidivism.

However, upon further analysis Hanson, Harris, Scott and Helmus (2007) indicated that only seven of the 16 items showed significant linear relationships to all recidivism categories (Negative Social Influences, AUC = .59; Hostility Toward Women., AUC = .58; Rejection/Loneliness, AUC = .60; Lack of Concern for Others,

AUC = .58; Lack of Cooperation with Supervision, AUC = .58; Impulsive Acts, AUC = .64; and Poor Cognitive Problem-Solving, AUC = .60). Three, of the items showed a significant linear relationship with the category of recidivism they were designed to predict: Sexual Pre-Occupations and Sex as Coping were significantly related to sexual recidivism, (AUCøs of .58 and .62 respectively) and Negative Emotion/Hostility was related to violent (AUC = .57) and general recidivism (AUC = .57). The ±Lovers/Intimate Partnersø item produced a significant but non-linear relationship to relevant outcomes (AUC = .59). None of the attitude items were significant to sexual offence recidivism, (sexual entitlement, AUC = .54; attitudes tolerant of rape, AUC = .54; attitudes tolerant of adult-child sex, AUC = 4.7), which is contrary to the theory and demonstrates the necessity for measures to be formally validated rather than simply empirically informed.

In addition, a 0.05 level of rejection was employed in Hanson, Harris, Scott and Helmusø, (2007) evaluation, which is acceptable. However, to further control the probability of a Type 1 error occurring it would be interesting to demonstrate the performance of the Stable-2000 at the more conservative level of rejection (0.01), particularly as the AUCøs at the 0.05 level are only moderate. Further, the predictive validity of the Stable-2000 remains somewhat premature and confirmatory analyses are required. However, the predictive validity presented may be affected because as with all supervisions the higher the risk of the offender then the closer the monitoring therefore providing opportunity less to offend and any offences committed more likely to be recorded.

Concurrent Validity

With no professional manual and no studies to date having provided correlation information on the Stable-200 and associated measures, its concurrent validity has not been firmly established. Future validations of the Stable-2000 should consider correlating the measure with dynamic risk assessments such as the Initial Deviance Assessment (IDA) (Thornton, 2002) and the Sex Offender Treatment Evaluation Project (STEP) Test Battery (Beech, Fisher, & Beckett, 1999). Nonetheless, preliminary item comparisons can be made with other dynamic assessments, which indicate that the Stable-2000 provides meaningful contributions to the risk assessment of sexual offenders. Furthermore, Kline (1986) states that for a measure to practically useful it needs to offer something that existing measures lack. In terms of risk assessment as a whole, the Stable-2000 offers a dynamic aspect, which static measures ignore. In comparison with existing dynamic measures, the Stable-2000 considers response to supervision, which has been consistently demonstrated as an important dynamic risk factor (e.g. Hanson & Harris, 2000; Quinsey, Rice, Harris, & Cormier, 1998; Zamble & Quinsey, 1997), whereas the IDA does not. In addition, it considers all sexual offenders whereas the STEP test battery was designed for use with adult male child abusers, and has not been validated for use with other populations.

Content Validity

The Stable-2000 has not been validated using an official measure of content validity, such as Lawshe (1975), who proposed that each item should be rated on its necessity to the performance of the construct by a panel of raters. Lawshe (1975) asserts that if more than half the raters agree on an itemsø necessity the measure can claim at least some content validity. However, the Stable-2000 can boast content validity in that it has evolved out of the extensive research area and theoretical background of sex offender risk assessment. It appears to consider all aspects of dynamic risk assessment. In addition, the authors are currently in the process of combining this into an integrated assessment considering areas of static, stable dynamic and acute dynamic risk.

Construct Validity

Due to the Stable-2000 being a relatively new assessment and the limited information pertaining to this it is difficult to assess the construct validity of the measure. High scores have been related to recidivism in Hanson, Harris, Scott and Helmusø (2007) study, which has been demonstrated via predictive validity analyses. To date, it seems that the Stable-2000 scores have not been related to any other relevant constructs or research areas. As the measure matures, it is hoped that the research and evidence based pertaining to this will increase.

NORMATIVE DATA

The importance of a professional manual is reiterated here for two reasons. Firstly, the Stable-2000 has been officially employed in many regions of Canada and in some establishments in the U.K. without established norms being published. Secondly, although a representative sample has been used in the subsequent validation study, clearly presented normative data is not available. The sample consisted of 805 adult sexual offenders over 16 different jurisdictions in Canada. The study does detail recidivism rates of these offenders. It also appropriately considers sex offenders not to be a homogenous group and thus identifies recidivism rates of rapists and child abusers separately. It is evident that clearly defined normative data is necessary. In addition, it would be useful to report normative data for U.K. samples.

CONCLUSIONS

The most salient criticism of the Stable-2000 is the lack of a professional manual and that is has been fully implemented in some jurisdictions in Canada and informally implemented in a number of settings in the U.K. without formal validation. In addition, the subsequent validity study has evidenced that some of the items in the Stable-2000 are in fact not predictive of future recidivism and many other areas of validity are neglected. Therefore, further highlighting the pervasive need for validation of this measure. However, the authors are currently in the process of revising the measure on the basis of the validation study which will result in the production of the Stable-2007. Furthermore, they are investigating the validity of a combined assessment incorporating the Static-99, Stable-2007, and acute-2007. Overall, the measure appears to be promising and there is clearly a requirement for such a measure within both the clinical and research fields. With further validation, a practical, reliable, and valid risk assessment measure is expected to emerge.

CHAPTER 5

Thesis Discussion

DISCUSSION

Risk assessment of sexual offenders is an important topic for researchers and clinicians working with sexual offenders, predominantly because sexual offending has significant impacts on the victims and wider society. As such, accurate risk assessments are essential, particularly has they have considerable impacts on the management of offenders. The primary of aim of this thesis was to examine developments in sexual offender risk assessment, with a focus on the possible relationships between key developmental variables, dynamic risk variables, and personality disorders. The thesis had a secondary aim to examine whether any of the risk related variables (key developmental variables, personality disorder, and dynamic factors) were related to treatment attrition.

A narrative review was presented in Chapter 1, which aimed to discuss and draw together four broad topic areas; dynamic risk factors, personality disorder, key developmental variables, and treatment adherence. A narrative review was presented here as the topic areas were too broad to compile a single systematic review. However, systematic search strategies were used in order to increase the validity of the review. The review demonstrates the significant advances in sex offender risk assessment in recent decades and describes advances in actuarial and dynamic risk assessment, with a discussion of risk variables. Detailed within the review are personality disorder and psychopathy, with particular reference to their relevance to sexual offending and risk. Key developmental or background variables are considered as risk factors and recent research querying their relationship to personality disorder and dynamic risk factors is presented. The review also considers treatment adherence

and how the risk variables are related to and possibly increase or predict risk. The review concludes that the evidence is there to suggest that further research in these areas is certainly warranted but at this time no firm conclusions can be drawn.

Chapter 2 presented an empirical study, seeking to go some way to alleviating the gap in the literature identified in the narrative review. Data was gathered from the Challenge Project, a community sex offender treatment programme, and information on 106 sexual offenders was available. The primary aims of the research were to investigate associations between items on the Stable-2007, key developmental variables, and personality disorder on the MCMI-III and to investigate the predictive value of these variables to failure on the treatment programme. A number of statistical analyses were performed. Chi-squared and Fishers Exact analyses revealed some significant associates between the variables. Low sample sizes were suggested to affect the results pertaining to failure on the programme as no significant associations were observed. Power analysis was conducted which suggested that increased sample sizes would make the results more promising for a number of variables; definite presence of PD, paranoid PD, definite presence of Cluster A PD, sadistic PD, and probable presence of Cluster B PD.

Unsurprisingly, from the results in Chapter two a relationship was found between social rejection and Cluster B personality disorder, Cluster C and Paranoid personality disorder. Individuals diagnosed with a personality disorder of any type would reasonably be expected to have some deficits in social interactions and combined with the stigma of such a label deems this an unsurprising result. Social rejection was also associated with emotionally/physically neglected. It may be that the skills required to

interact in social settings are suppressed under these childhood circumstances. Sex as coping was associated with a number of personality disorders including Histrionic and with one or more childhood traumas. Lack of concern for others was associated with dependant personality disorder and physical abuse and childhood traumas. This lack of concern may be developed from a negative view of others formed in childhood. The relationship to dependant personality disorder is seemingly less clear. However, it could perhaps be understood as although a negative view of others is developed so is a negative view of self and this confusion leads to a dependant style. Lack of significant social influences and relationship instability was again unsurprisingly associated with emotional and physical neglect, likely demonstrating a difficulty forming meaningful adult relationships due to the rejection experienced in childhood. Emotional identification with children was associated with sexual abuse, highlighting that abuse in childhood can lead to viewing children as emotionally safer alliances. Deviant sexual preferences were also associated with sexual abuse demonstrating the cognitive distortions formed during development.

In general, the results perhaps indicate a somewhat linear relationship from disruptions and abuse in childhood resulting in a skills deficit, to the development of personality disorder, leading and corresponding to difficulties within adult life. Therefore, this supports the notion that developmental variables, personality disorder and the Stable-2007 measure some underlying construct or developmental patterns.

A number of limitations with the empirical research were outlined, of which the sample size was key. Other methodological flaws were discussed focusing on data collection issues such as; missing information, relying on memory of other professionals, and conducting the Stable-2007 assessments from file information. It was concluded that the research demonstrates that this is a valid research topic. It provided weight to an emerging theory that there are associations between what we are measuring when we assess dynamic risk factors, key developmental variables, and personality disorder. However, it was apparent that more research in this is area is needed.

A single case study of Nigel, an adult rapist, was presented in Chapter 3, which provided a practical example of how risk issues can affect the management of sexual offenders. The case study was limited as the author was not able to complete the intervention and no post intervention measures were available at the time of writing. However, a detailed psychological assessment, risk assessment, formulation, and intervention plan (the Challenge Programme) was presented. The assessment revealed that Nigel was high risk on static measures and moderate risk on dynamic measures. He was also deemed to have high psychopathic tendencies. Nigel, therefore, provided a good example of a high risk sexual offender who is challenging in some way (psychopathy) and was relevant for this thesis. Two formulation hypotheses were presented in order to inform treatment plans. It was suggested that he may either have a dominant deviant sexual preference or a dominant antisocial and psychopathic personality. It was suggested that the Challenge Programme was the best option for Nigel in terms of community based treatments, although it was inconclusive of how he would fare in this given the lack of previous interventions. The biggest obstacle to treatment was deemed to be associated with psychopathy. This case study demonstrates the importance of individual assessments of sexual offenders whilst allowing for static and categorised risk assessments. It also demonstrates the need for responsivity issues, i.e. responding to individual clients needs, to be considered when a treatment programme is offered.

Chapter 1 of this thesis provided some evidence for the ubiquitous need for accurate, valid, and reliable risk assessments. However, Chapter 4 aimed to present a specific example of this relevant to the continuing development of dynamic risk assessments for sexual offenders. Specifically, Chapter 4 presented a critique of the Stable-2000 (Hanson & Harris, 2000) a dynamic risk assessment for child molesters and rapists. Within this Chapter, an overview of the measure was presented followed by a discussion of its psychometric properties; including various reliability and validity. In terms of reliability, the measure had good internal consistency. However, there was no information on test-retest reliability and the inter-rater reliability was questionable. In terms of validity, face validity, and content validity were deemed to be good, concurrent validity looked positive, and there was at least moderate predictive validity. However, there was no information on construct validity. The main criticism of the measure was that it was under-researched and there was a lack of a professional manual. The Stable-2000 has been updated and currently the Stable-2007 is available, which was used in the empirical research of this thesis. As such, it is hoped that future research can concentrate on validating the Stable-2007.

Conclusions

The field of sexual offender risk assessment and its advances in terms dynamic risk predictors, personality, and key developmental variables were examined in this thesis. The thesis has clearly demonstrated that over recent decades, sexual offender risk assessment has made many advances. Indeed, it provides some confirmation to

Bontags (1996) comment that there will be a forth generation risk assessment in the future. It is clear that dynamic risk assessments and the idea of combining static, stable dynamic, and acute dynamic assessments for a comprehensive assessment goes some way to support this. In addition, Craissatiøs work and the review and research presented in this thesis have demonstrated that the development of risk assessments considering background variables is an important development. The thesis has primarily demonstrated that the empirical research presented is preliminary and further research in this area needs to be conducted with more rigorous methodological considerations. However, the thesis gives good weight to hypotheses that key developmental variables, personality disorder, and dynamic risk factors are associated, perhaps by measuring some similar underlying constructs. The issue of treatment compliance was also examined throughout the thesis. Unfortunately, the empirical paper could draw no firm conclusions and further research is suggested with a larger sample size. However, the review and the case study identified that responsivity issues in treatment are important to minimise non-compliance. The future in terms of sexual offender risk assessment is a broad and ever developing one. It seems of importance to be considering holistic and comprehensive risk assessments taking care not to ignore background and individual factors.

REFERENCES

Alexander, M. (1999). Sexual offender treatment efficacy revisited. *Sexual Abuse: A Journal of Research and Treatment*, 11, 101-116.

Allan, A., Dawson, D., and Allan, M.M. (2006). Prediction of the risk of male sexual offending in Australia. *Australian Psychologist*, 41, 60-68.

Andrews, D.A. (1980). Some experimental investigations of the principles of differential association through deliberate manipulations of the structure of service systems. *American Sociological Review*, 45, 448-462.

Andrews, D.A., Bonta, J., and Hoge, R.D. (1990). Classification for effective rehabilitation: Rediscovering psychology. *Criminal Justice and Behaviour*, 17, 19-52.

Andrews, D. A., & Bonta, J. (1994). *The Psychology of Criminal Conduct*. Cincinnati, OH: Anderson Press.

Andrews, D. A., & Bonta, J. (1998). *The Psychology of Criminal Conduct (2nd Ed.)*. Cincinnati, OH: Anderson Press.

Andrews DA, Bonta J. (2003). *The Psychology of Criminal Conduct* (3rd Ed.). Cincinnati, Ohio, Anderson Publishing.

Andrews, D.A. and Dowden, C. (2006). Risk principle of case classification in correctional treatment. *International Journal of Offender Therapy and Comparative Criminology*, 50, 88-100.

Andrews, D. A., Dowden, C., & Gendreau, P. (1999). Clinically relevant and psychologically informed approaches to reduced re-offending: A meta-analytic study of human service, risk, need, responsivity, and other concerns in justice contexts. Unpublished manuscript, Carleton University, Ottawa, ON.

Bakker, L., Hudson, S., Wales, D., and Riley, D. (1998). And there was light... Evaluating the Kia Marama Programme for New Zealand Sex Offenders Against Children, Justice Department, Christchurch.

Barbaree, H.E. (1997). Evaluating treatment efficacy with sexual offenders: The insensitivity of recidivism studies to treatment effects. Sexual Abuse: A Journal of Research and Treatment, 9, 111-128.

Barbaree, H.E. Langton, C.M. and Peacock E.J. (2006). Different Actuarial Risk Measures Produce Different Risk Rankings for Sexual Offenders. *Sexual Abuse: A Journal of Research and Treatment*, 18, 423-440.

Barbaree H.E. and Marshall, W.L. (1989). Erectile responses among heterosexual child molesters, father-daughter incest offenders, and matched non-offenders; five distinct age preference profiles. *Canadian Journal of Behavioural Science*, 21, 70-82.

Barbaree, H.E., Seto, M.C., Langton, C.M and Peacock, E.J. (2001). Evaluating the Predictive Accuracy of Six Risk Assessment Instruments for Adult Sex Offenders. *Criminal Justice and Behaviour*, 28, 490-521.

Bard, L. A., Carter, D. L., Cerce, D. D., Knight, R. A., Rosenberg, R., & Schneider, B. (1987). A descriptive study of child molesters developmental clinical and criminal characteristics. *Behavioral Sciences and the Law*, 5,2036220.

Beech, A.R., Fisher, D & Beckett, R.C. (1999). An Evaluation of The Prison Sex Offender Treatment Programme. Home Office Occasional Paper. London: Home Office.

www.homeoffice.gov.uk/rds/pdfs/occ-step3.pdf

Beech, A.R., Fisher, D. & Thornton, D. (2003).Beech, A.R., Fisher, D.D., and Thornton, D. (2003). Risk assessment of sex offenders. *Professional Psychology, Research, and Practice*, 34, 339-352.

Beech, A., Friendship, C., Erikson, M., and Hanson, K.R. (2002). The Relationship Between Static and Dynamic Risk Factors and Reconviction in a Sample of U.K. Child Abusers. Sexual Abuse: *A Journal of Research and Treatment*, 14, 155-167.

Beech, A., Oliver, C., Fisher, D., and Beckett, R. (2005). STEP 4: The Sex Offender Treatment Programme in prison: addressing the needs of rapists and sexual

murderers. Available electronically from the Home Office website (www.homeoffice.gov.uk/rds).

Beech, A.R., and Ward, T. (2003). The integration of etiology and risk in sex offenders: A theoretical framework. *Aggression and Violent Behaviour*, 10, 31-63.

Berger, P., Berner, W., Bolterauer, J., Gutierrez, K., and Berger, K. (1999). Sadistic personality disorder in sex offenders: relationship to antisocial personality disorder and sexual sadism. *Journal of Personality Disorder*, 13, 175-86.

Bonta, J. (1996). Risk-needs assessment and treatment. In: A.T. Harland (ed.), *Choosing Correctional Options That Work*. Thousand Oaks, CA, Sage, pp 18-32.

Bonta, J., and Andrews, D.A. (2007). Risk-Need-Responsivity Model for Offender Assessment and Rehabilitation. Available at: http://ocipep-bpiepc.gc.ca/res/cor/rep/_fl/Risk_Need_2007-06_e.pdf.

Borchard, B., Gnoth, A., and Schulz, W. (2003) Personality disorders and "psychopathy" in sex offenders imprisoned in forensic-psychiatric hospitals--SKID-II- and PCL-R-results in patients with impulse control disorder and paraphilia. *Psychiatrische Praxis*, 30, 133-138.

Bowlby, J. (1977). The making and breaking of affectional bonds. II. Some principles of psychotherapy: The fiftieth Maudsley Lecture. *The British Journal of Psychiatry*, 130, 421-431.

Bowlby, J. (1979). On knowing what you are not supposed to know and feeling what you are not supposed to feel. *Canadian Journal of Psychiatry*, 24, 403-408.

Brown, S.L., and Forth, A.E. (1997). Psychopathy and sexual assault: Static risk factors, emotional precursors, and rapist subtypes. *Journal of Consulting and Clinical Psychology*, 65, 8486857.

Bumby, K.M. (1996). Assessing the cognitive distortions of child molesters and rapists: Development and validation of the MOLEST and RAPE scales. <u>Sexual Abuse:</u> A <u>Journal of Research and Treatment</u>, 8, 37-54.

Buss, A.H., and Durkee, A. (1957). An inventory for assessing different kinds of hostility. *Journal of Consulting Psychology*, 21, 343-349.

Cattell, R. B., & Klein, T. W. (1975). A check on hypothetical personality structures and their interpretations at 14-16 years in T-data. *British Journal of Psychology*, 66, 131-151.

Chantry, K. & Craig, R. J. (1994). MCMI typologies of criminal sexual offenders. *Sexual Addiction and Compulsivity*, 1, 2156226.

Cleckley, H. (1982). The Mask of Sanity. Revised Edition. Mosby Medical Library

Cooke, D.J. (1995). Psychopathic disturbance in the Scottish prison population: The cross-cultural generalisability of the Hare Psychopathy Checklist. *Psychology, Crime, and Law,* 2, 101-108.

Collins, J. A., & Fauser, B. C. (2005). Balancing the strengths of systematic and narrative reviews. *Human Reproductive Update*, 11, 103–104.

Craig, L.A., and Beech, A. (2009). Best practice in conducting actuarial risk assessments with adult sexual offenders. *Journal of Sexual Aggression*, 15, 193-211.

Craig, L. A., Beech, A. R., & Browne, K. D. (2006). Evaluating the predictive accuracy of sex offender risk assessment measures on UK a Samples: A cross-validation of the risk matrix 2000 scales. *Sex offender treatment*, 1, 1-16.

Craig, L.A., Browne, K. D., and Stringer, I. (2003a) Risk Scales and Factors Predictive Of Sexual Offence Recidivism. *Trauma, Violence, & Abuse,* 4, 45-69.

Craig, L.A., Browne, K.D., Stringer, I., Beech, A.R. (2005). Sexual recidivism: A review of static, dynamic and actuarial risk predictors, *Journal of Sexual Aggression*, 11, 65-84.

Craissati, J., and Beech, A. (2001). Attrition in a community treatment program for child sexual abusers. *Journal of Interpersonal Violence*, 16, 205-221.

Craissati, J., and Beech, A. (2002). *Risk prediction and failure in a complete geographical sample of sex offenders*. Unpublished manuscript.

Craissati, J., and Beech, A. (2003). A review of dynamic variables and their relationship to risk prediction in sex offenders. *Journal of Sexual Aggression*, 9, 41-55.

Craissati, J., and Beech, A. (2004). The characteristics of a geographical sample of convicted rapists: Sexual victimisation and compliance in comparison to child molesters. *Journal of Interpersonal Violence*, 19, 371-388.

Craissati, J., & Beech, A. (2006). The role of key developmental variables in identifying sex offenders likely to fail in the community: An enhanced risk prediction model. *Child Abuse & Neglect*, 30, 327-339.

Craissati, J., Falla, S., McClurg, G., and Beech, A.R. (1998). Risk, reconviction rates and pro-offending attitudes for child molesters in a complete geographical area of London. *Journal of Sexual Aggression*, 8, 1, 22 - 38

Craissati, J., and McClurg, G. (1996). The challenge project: Perpetrators of child sexual abuse in south east London. *Child Abuse & Neglect*, 20, 1067ó1077.

Craissati, J., McClurg, G., & Browne, K. (2002a). Characteristics of perpetrators of child sexual abuse who have been sexually victimized as children. *Child Abuse* & *Neglect*, 14, 225-239.

Craissati, J., McClurg, G., & Browne, K. (2002b). The parental bonding experiences of sex offenders: a comparison between child molesters and rapists. *Child Abuse & Neglect*, 26, 9096921.

Doren, M. D., and Yates, P.M. (2008). Effectiveness of Sex Offender Treatment for Psychopathic Sexual Offenders. *International Journal of Offender Therapy and Comparative Criminology*, 52, 234, 245.

DøSilva, K., Duggan, C., and McCarthy, L. (2004). Does Treatment Really Make Psychopaths Worse? A Review of the Evidence. *Journal of Personality Disorders*, 18, 163-177.

Dutton, D. G. (1995a). Male abusiveness in intimate relationships. *Clinical Psychology Review*, 15, 567-581.

Farrington, D. P. (2006b) Family background and psychopathy. In Patrick, C. J. (Ed.) *Handbook of Psychopathy*. New York: Guilford Press (pp. 229-250).

Kroner, D. G., & *Forth*, A. E. (*1995*). The Toronto Alexithymia Scale with incarcerated offenders. *Personality and Individual Differences*, 19, 625-634.

Friendship, C., Mann, R.E., and Beech, A.R. (2003). Evaluation of a National Prison-based Treatment Program for Sexual Offenders in England and Wales. *Journal of Interpersonal Violence*, 18, 744-759

Friendship, C. and Thornton, D. (2001). Sexual reconviction for sexual offenders discharged from prison in England and Wales: Implications for evaluating treatment. *British Journal of Criminology*, 41, 285-292.

Finkelhor, D. (1984) *Child Sexual Abuse: New Theory and Research*. New York: Free Press.

Ginsburg, J.I.D., Mann, R.E., Rotgers, F., and Weekes, J.R. (2002). Motivational Interviewing with Criminal Justice Populations. In: W. R. Miller and S. Rollnick (Eds.), *Motivational Interviewing: Preparing People for Change-2nd edition*. New York, NY: Guilford Press.

Guilford, J.P. (1956). *The Structure of Intellect*. Psychological Bulletin 53(4), 267-293.

Grossman, L.S., Martis, B., and Fitchtner, C.G. (1999). Are sex offenders treatable? A research overview. *Psychiatric Services*, 50, 349-361.

Hall, G.C.N. (1995). Sexual offender recidivism revisited: A meta-analysis of recent treatment studies. *Journal of Consulting and Clinical Psychology*, 63, 802-809.

Hall, G. C. N., & Hirschman, R. (1992). Sexual aggression against children: A conceptual perspective of etiology. *Criminal Justice and Behaviour*, 19, 8-23.

Hanson, R.K. (2000). Sex offender risk assessment. In C.R. Hollin (Eds.), *Handbook of Offender Assessment*. (pp.87-95). Bath. Bookcraft Ltd.

Hanson, R. K., Gordon, A., Harris, A., Marques, J., Murphy, W., Quinsey, V., & Seto, M. (2002). The effectiveness of treatment for sexual offenders. *Sexual Abuse: A Journal of Research and Treatment*, 14,169-194.

Hanson, R.K., Bourgon, G., Helmus, L., and Hodgson, S. (2009). The principles of effective correlational treatment also apply to sexual offenders: A meta-analysis. *Criminal Justice and Behaviour*, 36, 865-891.

Hanson, R.K., & Bussière, M.T. (1998). Predicting relapse: A meta-analysis of sexual offender recidivism studies. *Journal of Consulting and Clinical Psychology*, 66, 2, 348-362.

Hanson, R.K., Broom, I., and Stephenson, M. (2004). Evaluating community sex offender treatment programs: A 12-year follow-up of 724 offenders. *Canadian Journal of Behavioural Science*, 36, 87-96.

Hanson, R. K., & Harris, A. J. R. (2000). Where should we intervene: Dynamic predictors of sexual assault recidivism. *Criminal Justice and Behavior*, 27, 6635.

Hanson, R.K., and Harris, A. (2001). *The Sex Offender Need Assessment Rating (SONAR): A method for measuring change in risk levels 2000-2001*, Ottowa, Department of the Solicitor General of Canada.

Hanson, R. K., Harris, A. J. R., Scott, T., & Helmus, L. (2007). Assessing the risk of sexual offenders on community supervision: The dynamic supervision project. Ottawa, ON: Public Safety and Emergency Preparedness Canada.

Hanson, R.K., and Morton-Bourgon, K. (2004). *Predictors of sexual recidivism: An updated meta-analysis* (Corrections Research User Report No. 2004-02). Ottawa, Ontario: Public Safety of Canada.

Hanson R.K., and Morton-Bourgon, K. E. (2005). The characteristics of persistent sexual offenders: A meta-analysis of recidivism studies. *Journal of Consulting and Clinical Psychology*, 73, 1154-1163.

Hanson, R. K., Morton, K.E., and Harris, A.J.R. (2003). Sexual offender recidivism risk: What we know and what we need to know. Annals of the New York Academy of Science, 989, 154-166.

Hanson, R. K., & Thornton, D. (1999). *Static-99: Improving actuarial risk assessments for sex offenders*. Ottawa, Ontario, Canada: Department of the Solicitor General.

Hanson, R.K., & Thornton, D. (2000). Improving risk assessments for sex offenders: A comparison of three actuarial scales. *Law and Human Behaviour*, 24, 1, 119-136.

Hare, R. (1991). *Manual for the Revised Psychopathy Checklist*. Toronto, Canada: Multi-Health Systems.

Hare, R. D. (1996a). A clinical construct whose time has come. *Criminal Justice and Behavior*, 23, 25654.

Hare, R. D. (1996b). Psychopathy and antisocial personality disorder: A case of diagnostic confusion. *Psychiatric Times*, 39640.

Hare, R. D (1998) Psychopathy, affect and behaviour. In: D. Cooke, A. Forth and R. Hare (Eds.), *Psychopathy: Theory, Research and Implications for Society*, pp. 1056-139. Dordrecht: Kluwer.

Hare, R. D., Clark, D., Grann, M., & Thornton, D. (2000). <u>Psychopathy and the predictive validity of the PCL-R: An international perspective.</u> *Behavioral Sciences and the Law*, 18, 623-645.

Harkins, L. and Beech, A.R. (2007). Measurement of the effectiveness of sex offender treatment. *Aggression and Violent Behaviour*, 12, 36-44.

Hart, S. D., Cox, D. N., & Hare, R. D. (1995). *The Hare Psychopathy Checklist: Screening Version.* Toronto, ON: Multi-Health Systems.

Hart, S. D., Forth, A. E., & Hare, R. D. (1991). Psychopathy and the Millon Clinical Multiaxial Inventory-II. *Journal of Personality Disorders*, *5*, 318-327.

Hart, S. D., & Hare, R. D. (1997). Psychopathy: Assessment and association with criminal conduct. In D. Stoff, J. Breiling, & J. Maser (Eds.), *Handbook of antisocial behavior* (pp. 22-35). New York, NY: Wiley & Sons.

Harris, A. J. R., & Hanson, R. K. (2004). *Sex offender recidivism: A simple question* (No. 2004-03). Ottawa: Public Safety and Emergency Preparedness Canada.

Harris, G. T., Rice, M. E., & Cormier, C. A. (2002). Prospective replication of the Violence Risk Appraisal Guide in predicting violent recidivism among forensic patients. *Law and Human Behavior*, 26, 3776394.

Harris, G. T., Rice, M.E., Quinsey, V.L., Lalumiere, M.L., Boer, D., Lang, C. (2003). A Multisite Comparison of Actuarial Risk Instruments for Sex Offenders, *Psychological Assessment*, 3, 413-425.

Hildebrand, M., de Ruiter, C., and de Vogel, V. (2004). Psychopathy and sexual deviance in treated rapists: Association with sexual and nonsexual recidivism. *Sexual Abuse: A Journal of Research and Treatment*, 16, 1-24.

Hill, C. D., Neumann, C. S., and Rogers, R. (2004). Confirmatory factor analysis of the Psychopathy Checklist: Screening Version in offenders with Axis I disorders. *Psychological Assessment*, 16, 90695.

Hobson, J., Shine, J., and Roberts, R. (2000). How do psychopaths behave in a prison therapeutic community? *Psychology, Crime & Law,* 6, 139 ó 154.

Hood, R., Shute, S., Feilzer, M., and Wilcox, A. (2002). Sex Offenders Emerging from Long-Term Imprisonment. A Study of Their Long-term Reconviction Rates and of Parole Board Members' Judgements of Their Risk. **The British Journal of Criminology**, 42, 371-394

Howells, K., and Day, D. (2003). Readiness for anger management: clinical and theoretical issues. *Clinical Psychology Review*, 23, 319-337.

Hudson, S.M., Wales, D.S., Bakker, L., Ward, T. (2004). Dynamic risk factors: The Kia Marama evaluation. *Sexual Abuse: A Journal of Research and Treatment*, 14, 103, 119.

Hudson, S.M., and Ward, T. (1997). Intimacy, Loneliness, and Attachment Style in Sexual Offenders. *Journal of Interpersonal Violence*, 12, 323-339.

Kiehl, K.A., Hare, R.D., McDonald, J.J., and Brink, J. (1999) Semantic and affective processing in psychopaths: An event-related potential (ERP) study. *Psychophysiology*, 36, 765-774.

Kemshall, H. (1996a) Reviewing Risk: A review of research on the assessment and management of risk and dangerousness: implications for policy and practice in the Probation Service. A report for the Home Office Research and Statistics Directorate. London: Home Office.

Kennedy, H.G and Grubin, D.H. (1992). Patterns of denial in sex offenders. *Psychological Medicine*, 22, 191-196.

Kenworthy, T., Adams, C.E., Brooks-Gordon, B., and Fenton, M. (2004). *Psychological interventions for those who have sexually offended or are at risk of offending* (CD004858; Cochrane Database of Systematic Reviews, Issue 3). Chichester, UK: John Wiley & Sons.

Kline, P. (1986). A handbook of test construction: Introduction to psychometric design. Methuen, New York.

Kingston, D.A., Yates, P.M., Firestone, P., Babchishin, K., and Bradford, J. (2008). Long term predictive validity of the Risk Matrix 2000: A comparison with the Static-99 and the Sex Offender Risk Appraisal Guide. *Sexual Abuse: A Journal of Research and Treatment*, 20, 466-484.

Knight, R., and Sims-Knight, J.E. (2003). The development of antecedents of sexual coercion against women: testing alternative hypotheses with structural equation modelling. *Annals of the New York Academy of Sciences*, 989, 72-85.

Kosson, D.S., Smith, S.S., and Newman, J.P. (1990). Evaluating the Construct Validity of Psychopathy in Black and White Male Inmates: Three Preliminary Studies. *Journal of Abnormal Psychology*, 3, 250-259.

Langstrom, N., and Grann, M. (2000). Risk for criminal recidivism among young sex offenders. *Journal of Interpersonal Violence*, 15, 855-871.

Langstom, N., Sjostegt, G., and Grann, M. (2004). Psychiatric disorders and recidivism in sexual offenders. *Sexual Abuse: A Journal of Research and Treatment*, 16, 139-150.

Lawshe, G.H. (1975). A quantitative approach to content validity. *Personnel Psychology*, 28, 563-575.

Looman, J., Abracen, J., Serin, R., and Marquis, P. (2005). Psychopathy, treatment change, and recidivism in high-risk, high-need sexual offenders. *Journal of Interpersonal Violence*, 20, 549-568.

Looman, J., Dickie, I., and Abracen, J. (2005). Responsivity issues in the treatment of sexual offenders. *Trauma, Violence, & Abuse,* 6, 330-353.

Lynam, D.R., Caspi, A., Moffitt, T.E., Loeber, R., and Stouthamer-Loeber, M. (2007). The content validity of juvenile Psychopathy: An empirical examination, *Psychological Assessment*, 19, 363-367.

Malamuth, N.M. (2003). Criminal and noncriminal sexual aggressors. <u>Annals of the</u> New York Academy of Sciences, 989, 33 ó 58.

Mann, R.E., Ginsburg, J.I.D., and Weekes, J.R. (2002). Motivational interviewing with offenders. In: M. McMurran (Ed.), *Motivating Offenders to Change: A Guide to Enhancing Engagement in Therapy*. New York: John Wiley & Sons.

Marques, J.K., Day, D.M., Nelson, C., and West, M.A. (1994). Effects of cognitive behavioural treatment on sex offender recidivism preliminary results of a longitudinal study. *Criminal Justice and Behavior*, 21, 28-54.

Marshall, W.L. and Barbaree, H.E. (1990). An integrated theory of sexual offending. In: W.L. Marshall, D.R. Laws and H.E. Barbaree (Eds.), *Handbook of sexual assault: Issues, theories, and treatment of the offender.* (pp. 257275). New York: Plenum Press.

Marshall, W. L., Hudson, S. N., and Hodkinson, S. (1993). The importance of attachment bonds and the development of juvenile sex offending. In: H. E. Barbaree, W. L. Marshall, and S. M. Hudson (Eds.), *The Juvenile Sex Offender* (pp. 164-181). New York: Guilford.

McDermot, P.A., Alterman, A.I., Cacciola, J.S., Rutherford, M.J., Newman, J.P., and Mulholland, E.M. (2000). Generality of Psychopathy Checklist-Revised factors over prisoners and substance dependent patients. *Journal of Consulting and Clinical Psychology*, 68, 181-186.

Million, T., Millon, C., Davis, R., and Grossman, S. (1997). MCMI-III: manual, Minneapolis, MN: NCS Pearson

Neumann, C. S., Kosson, D. S., Forth, A. E., & Hare, R. D. (2006). <u>Factor structure of the Hare Psychopathy Checklist: Youth Version in incarcerated adolescents</u>. *Psychological Assessment*, 18, 142-154.

Nichols, H. R., & Molinder, I. (1984). *Multiphasic Sex Inventory Manual*. Tacoma, WA: Author.

Nunes, K. L., Firestone, P., Bradford, J. M., Greenberg, D. M., & Broom, I. (2002). A comparison of modified versions of the Static-99 and Sex Offender Risk Appraisal Guide. *Sexual Abuse: A Journal of Research and Treatment*, 14, 2536269.

Olver, M.E., and Wong, S.C.P. (2006). Psychopathy, sexual deviance, and recidivism among sex offenders. *Sex Abuse*, 18, 65-82.

Olver, M.E., Wong, S.C.P., Nicholaichuk, T.P. (2009). Outcome evaluation of a high intensity inpatient sex offender treatment program. *Journal of Interpersonal Violence*, 24, 522-536.

Pithers, W.D., Kashima, K.M., Cumming, G.F., & Beal, L.S. (1988). Relapse prevention: A method of enhancing maintenance of change in sex offenders. In A.C. Salter (Ed.), *Treating child sex offenders and victims* (pp. 131-170). London: Sage.

Porter, S., Fairweather, D., Drugge, J., Herve, H., Birt, A. R., & Boer, D. P. (2000). Profiles of psychopathy in incarcerated sexual offenders. *Criminal Justice & Behavior*, 27, 2166233.

Prins, H. (2005). Taking Chances: Risk assessment and management in a risk obsessed society. *Medicine, Science, and Law*, 45, 93-209.

Prochaska, J.O. and DiClemente, C.C. (1982). Transtheoretical therapy: Toward a more integrative model of change. *Psychotherapy: Theory, Research, and Practice*, 19, 276-288.

Proulx, J., Perreault C., and Ouimet, M. (1999). Pathways in the offending process of extrafamilial sexual child molestors. *Sexual abuse: A Journal of Research and Treatment*, 11, 117-129.

Quinsey, V.L. (1998). Treatment of sex offenders. In M. Tonry (Ed.), *The handbook of crime and punishment* (pp. 403-425). New York: Oxford ariality Press.

Quinsey, V.L., Lalumiere, M.L., Rice, M.E., & Harris, G.T. (1995). Predicting sexual offenses. In J.C. Campbell (Ed.), *Assessing dangerousness: Violence by sexual offenders, batterers, and child abusers* (pp. 114-137). Thousand Oaks, CA: Sage.

Quinsey, V.L., Rice, M.E., & Harris, G.T. (1995). Actuarial prediction of sexual recidivism. *Journal of Interpersonal Violence*, 10, 85-105.

Quinsey, V. L., Harris, G. T., Rice, M. E., and Cormier, C. (1998). *Violent Offenders: Appraising and Managing Risk*. Washington DC: American Psychological Association.

Rice, M.E., and Harris, G.T., (1997). Cross-validation of the violence risk appraisal guide for child molesters and rapists. *Law and Human Behaviour*, 21, 231-241.

Rice, M.E., Harris, G.T., & Quinsey, V.L. (1990). A follow-up of rapists assessed in a maximum security psychiatric facility. *Journal of Interpersonal Violence*, 5, 435-448.

Seto, M.C., and Barbaree, H.E. (1999). Psychopathy, treatment behaviour, and sex offender recidivism. *Journal of Interpersonal Violence*, 14, 1235-1248.

Seghorn, T.K., Boucher, R.J., and Prentsky, R.A. (1987). Childhood sexual abuse in the lives of sexually aggressive offenders. *Journal of American Academy of Child and Adolescent Psychiatry* 26, 262-267.

Schmucker, M., and Losel, F. (2008). Does sexual offender treatment work? A systematic review of outcome evaluations. *Psicothema*, 20, 10-19.

Sjostedt, G., and Landstrom, N. (2002). Assessment of Risk for Criminal Recidivism Among Rapists: A Comparison of Four Different Measures. *Psychology, Crime, and Law*, 8, 28-40

Smallbone, S.W., and Dadds, M.R. (1998). Childhood attachment and adult attachment in incarcerated male sex offenders. *Journal of Interpersonal Violence*, 13, 555-573.

Stephenson, M. (1991). A summary of an evaluation of the community sex offender program in the Pacific region. *Forum on Corrections Research*, 3, 25-30.

Thornton, D. (2000, March). *Structured risk assessment*. Paper presented at the Sinclair Seminars Conference on Sex Offender Re Offense Risk Prediction, Madison, WI. (Videotape available from www.sinclairseminars.com).

Thornton, D. (2002) Constructing and testing a framework for dynamic risk assessment. Sexual Abuse: A Journal of Research and Treatment, 14, 139-153.

Thornton, D., Mann, R., Webster, S., Blud, L., Travers, R., Friendship, C. et al. (2003). Distinguishing and combining risks for sexual and violent recidivism. In R. Prentky, E. Janus and M. Seto (Eds.), *Understanding and Managing Sexually*

Coercive Behaviour, Vol. 989 (pp. 225-235). New York: Annals of the New York Academy of Sciences.

Tierney, D.W., and McCabe, M.P. (2002). Motivation for behaviour change among sex offenders: A review of the literature. *Clinical Psychology Review*, 22, 113-129.

Viding, E., Blair, J. R., Moffitt, T. E., and Plomin, R. (2005). Psychopathic syndrome indexes strong genetic risk for antisocial behavior in 7yearolds. *Journal of Child Psychology and Psychiatry*, 46, 5926597.

Viding, E., Frick, P.J., Plomin, R. (2007). Aetiology of the relationship between callous-unemotional traits and conduct problems in childhood. The British Journal of Psychiatry, 190, 33-38.

Ward, T., Hudson, S.M., Johnston, W., and Marshall, W.L. (1997). Cognitive distortions in sex offenders: An integrative review. *Clinical Psychology Review*, 17, 497-507

Ward, T., Hudson, S.M., and Marshall, W.L. (1996). Attachment style in sex offenders: A preliminary study. *Journal of Sex Research*, 33, 17-26.

Ward, T., Louden, K., Hudson, S. and Marshall, W. L. (1995). A Descriptive Model of the Offence Chain in Child Molesters. *Journal of Interpersonal Violence*, 10, 453-473.

Ward, T., & Seigert, R. J. (2002). Towards a comprehensive theory of child sexual abuse: A theory knitting perspective. *Psychology, Crime, and Law,* 8, 319 ó 351.

Ward, T., & Stewart, C. A. (2003). The treatment of sexual offenders: Risk management and good lives. *Professional Psychology: Research and Practice*, 34, 353-360.

Watkins, B., and Bentovim, A. (1992). The sexual abuse of male children and adolescents: a review of current research. *Journal of Child Psychology and Psychiatry*, 33, 1976248.

Widom, C. (1989). Child abuse, neglect and adult behavior: research design and findings on criminality, violence and child abuse. *American Journal of Orthopsychiatry*, 59, 3556367.

Webster, C.D., Eaves, D., Douglas, K.S., and Winthrup, A. (1995). *The HCR-20 scheme. The assessment of dangerousness and risk.* Vancouver, Simon Fraser University and British Columbia Forensic Psychiatric Services Commission.

Wormith, J.S., Olver, M.E., Stevenson, H.E., & Girard L. (2007). The long-term prediction of offender recidivism using diagnostic, personality, and risk/need approaches to offender assessment. *Psychological Services*, 4, 287-305.

Wong, S., & Hare, R. D. (2005). *Guidelines for a Psychopathy Treatment Program*. Toronto, ON: Multi-Health Systems.

Zamble, E. & Quinsey, V.L. (1997). *The Criminal Recidivism Process*. New York, NY: Cambridge University Press.

APPENDICIES

<u>APPENDIX 1 - LIST OF COCHRANE REFERENCES</u>

R=Relevant, NR=Not Relevant

Clinical Trials (n=38):

(R=1, NR=37)

Annunziato, R.A., Timko, C.A., Crerand, C.E., Didie, E.R., Bellace, D.L., Phelan, S., Kerzhnerman, I., and Lowe, M.R. (2009). A randomized trial examining differential meal replacement adherence in a weight loss maintenance program after one-year follow-up. Eating behaviours, 2009, 10, 176-83. (NR)

Ball, J.R., Mitchell, P.B., Corry, J.C., Skillecorn, A., Smith, M., and Malhi, G.S. (2006). A randomized controlled trial of cognitive therapy for bipolar disorder: focus on long-term change. The Journal of clinical psychiatry, 67, 277-86. (NR)

Bancroft, J., Tennent, G., Loucas, K., and Cass, J. (1974). The control of deviant sexual behaviour by drugs. I. Behavioural changes following oestrogens and anti-androgens. The British journal of psychiatry: the journal of mental science, 125, 310-5. (NR)

Borduin, C.M., Schaeffer, C.M., and Heiblum, N. (2009). A randomized clinical trial of multisystemic therapy with juvenile sexual offenders: effects on youth social ecology and criminal activity. Journal of consulting and clinical psychology, 77, 26-37. (NR)

Braithwaite, R.L., Stephens, T.T., Treadwell, H.M., Braithwaite, K., and Conerly, R. (2005). Short-term impact of an HIV risk reduction intervention for soon-to-be released inmates in Georgia. Journal of health care for the poor and underserved, 16, 130-9. (NR)

Courneya, K.S., Friedenreich, C.M., Quinney, H.A., Fields, A.L., Jones, L.W., and Fairey, A.S. (2004). Predictors of adherence and contamination in a randomized trial of exercise in colorectal cancer survivors. Psycho-oncology, 13, 857-66. (NR)

Cowan, M.J., Freedland, K.E., Burg, M.M., Saab, P.G., Youngblood, M.E., Cornell, C.E., Powell, L.H., and Czajkowski, S.M. (2008). Predictors of treatment response for depression and inadequate social support--the ENRICHD randomized clinical trial. Psychotherapy and psychosomatics, 77, 27-37. (NR)

Doesschate, M.C., Bockting, C.L., Koeter, M.W., and Schene, A.H. (2009). Predictors of nonadherence to continuation and maintenance antidepressant medication in patients with remitted recurrent depression. The Journal of clinical psychiatry, 70, 63-9. (NR)

Gladis, M.M., Wadden, T.A., Vogt, R., Foster, G., Kuehnel, R.H., and Bartlett, S.J. (1998). Behavioral treatment of obese binge eaters: do they need different care? Journal of psychosomatic research, 44, 375-84. (NR)

Gopinath, S., Katon, W.J., Russo, J.E., and Ludman, E.J. (2007). Clinical factors associated with relapse in primary care patients with chronic or recurrent depression. Journal of affective disorders, 101, 57-63 (NR)

Grubin, D., Madsen, L., Parsons, S., Sosnowski, D., and Warberg, B. (2004). A prospective study of the impact of polygraphy on high-risk behaviors in adult sex offenders. Sexual abuse: a journal of research and treatment, 16, 209-22. (NR)

Henggeler, S.W., Letourneau, E.J., Chapman, J.E., Borduin, C.M., Schewe. P.A., and McCart, M.R. (2009). Mediators of change for multisystemic therapy with juvenile sexual offenders. Journal of consulting and clinical psychology, 77, 451-62. (NR)

Katon, W.J., Von Korff, M., Lin, E.H., Simon, G., Ludman, E., Russo, J., Ciechanowski, P., Walker, E., and Bush, T. (2004). The Pathways Study: a randomized trial of collaborative care in patients with diabetes and depression. Archives of general psychiatry, 61,1042-9. (NR)

Katon, W., Von Korff, M., Lin, E., Simon, G., Walker, E., Unützer, J., Bush, T., Russo, J., and Ludman, E. (1999). Stepped collaborative care for primary care patients with persistent symptoms of depression: a randomized trial. Archives of general psychiatry, 56, 1109-15. (NR)

Kleindienst, N., and Greil, W. (2004). Are illness concepts a powerful predictor of adherence to prophylactic treatment in bipolar disorder? The Journal of clinical psychiatry, 65, 966-74. (NR)

Kolla, N.J., Links, P.S., McMain, S., Streiner, D.L., Cardish, R., and Cook, M. (2009). Demonstrating adherence to guidelines for the treatment of patients with borderline personality disorder. Canadian journal of psychiatry. Revue canadienne de psychiatrie, 54, 181-9. (NR)

Lin, E.H., Simon, G.E., Katon, W.J., Russo, J.E., Von Korff, M., Bush, T.M., Ludman, E.J., and Walker, E.A. (1999). Can enhanced acute-phase treatment of depression improve long-term outcomes? A report of randomized trials in primary care. The American journal of psychiatry, 156, 643-5. (NR)

Linehan, M.M. (1998). Therapy for Borderline personality and drug abuse.1 51st Annual Meeting of the American Psychiatric Association. http://www.mrw.interscience.wiley.com/cochrane/clcentral/articles/379/CN-00282379/frame.html. (NR)

Lynch, D.J., Birk, T.J., Weaver, M.T., Gohara, A.F., Leighton, R.F., Repka, F.J., and Walsh, M.E. (1992). Adherence to exercise interventions in the treatment of hypercholesterolemia. Journal of behavioral medicine, 15, 365-77. (NR)

Mann, R.E., Webster, S.D., Schofield, C., and Marshall, W.L. (2004). Approach versus avoidance goals in relapse prevention with sexual offenders. Sexual abuse: A journal of research and treatment, 16, 65-75. (NR)

Marques, K., Day, D.M., Nelson, C., West, M.A. (1994). Effects of cognitive-behavioral treatment on sex offender recidivism: Preliminary results of a longitudinal study. Special Issue: The assessment and treatment of sex offenders. Criminal-Justice-and-Behavior, 21, 28-54. (R)

Mataix-Cols, D., Marks, I.M., Greist, J.H., Kobak, K.A., and Baer, L. (2002). Obsessive-compulsive symptom dimensions as predictors of compliance with and response to behaviour therapy: results from a controlled trial. Psychotherapy and psychosomatics, 71, 255-62. (NR)

Mavissakalian, M.R. (2003). Imipramine vs. sertraline in panic disorder: 24-week treatment completers. Annals of clinical psychiatry: official journal of the American Academy of Clinical Psychiatrists,15,171-80. (NR)

Miner, M.H., and Center, B. (2008). Improving the measurement of criminal sexual behavior: the application of randomized responding technique. Sexual abuse: A journal of research and treatment, 20, 88-101. (NR)

Mohr, D.C., Likosky, W., Bertagnolli, A., Goodkin, D.E., Van Der Wende, J., Dwyer, P., and Dick, L.P. (2000). Telephone-administered cognitive-behavioral therapy for the treatment of depressive symptoms in multiple sclerosis. Journal of consulting and clinical psychology, 68, 356-61. (NR)

Perahia, D.G., Quail, D., Gandhi, P., Walker, D.J., and Peveler, R.C. (2008) A randomized, controlled trial of duloxetine alone vs. duloxetine plus a telephone intervention in the treatment of depression. Journal of affective disorders, 108, 33-41. (NR)

Pettinati, H.M., Monterosso, J., Lipkin, C., and Volpicelli, J.R. (2003). Patient attitudes toward treatment predict attendance in clinical pharmacotherapy trials of alcohol and drug treatment. The American journal on addictions, 12, 324-35. (NR)

Pugh, R. (1983). An association between hostility and poor adherence to treatment in patients suffering from depression. The British journal of medical psychology, 56, 205-8. (NR)

Resick, P.A., Galovski, T.E., O'Brien, U.M., Scher, C.D., Clum, G.A., and Young-Xu, Y. (2008). A randomized clinical trial to dismantle components of cognitive processing therapy for posttraumatic stress disorder in female victims of interpersonal violence. Journal of consulting and clinical psychology, 76, 243-58. (NR)

Ruskin, P.E., Silver-Aylaian, M., Kling, M.A., Reed, S.A., Bradham, D.D., Hebel, J.R., Barrett, D., Knowles, F., and Hauser, P. (2004). Treatment outcomes in depression: comparison of remote treatment through telepsychiatry to in-person treatment. The American journal of psychiatry, 161, 1471-6. (NR)

Schaefer, M., Hinzpeter, A., Mohmand, A., Janssen, G., Pich, M., Schwaiger, M., Sarkar, R., Friebe, A., Heinz, A., Kluschke, M., Ziemer, M., Gutsche, J., Weich, V., Halangk, J., and Berg, T. (2007). Hepatitis C treatment in "difficult-to-treat" psychiatric patients with pegylated interferon-alpha and ribavirin: response and psychiatric side effects. Hepatology, 46, 991-8. (NR)

Shapiro, S.L., Oman, D., Thoresen, C.E., Plante, T.G., and Flinders, T. (2008). Cultivating mindfulness: effects on well-being. Journal of clinical psychology, 64, 840-62. (NR)

Shemesh, E., Koren-Michowitz, M., Yehuda, R., Milo-Cotter, O., Murdock, E., Vered, Z., Shneider, B.L., Gorman, J.M., and Cotter, G. (2006). Symptoms of posttraumatic stress disorder in patients who have had a myocardial infarction. Psychosomatics, 47, 231-9. (NR)

Simon, G.E., Katon, W., Rutter, C., VonKorff, M., Lin, E., Robinson, P., Bush, T., Walker, E.A., Ludman, E., and Russo, J. (1998). Impact of improved depression treatment in primary care on daily functioning and disability. Psychological medicine, 28, 693-701. (NR)

Stilley, C.S., Sereika, S., Muldoon, M.F., Ryan, C.M., and Dunbar-Jacob, J. (2004). Psychological and cognitive function: predictors of adherence with cholesterol lowering treatment. Annals of behavioral medicine: a publication of the Society of Behavioral Medicine, 27,117-24. (NR)

St Lawrence, J.S., Crosby, R.A., Belcher, L., Yazdani, N., and Brasfield, T.L. (1999). Sexual risk reduction and anger management interventions for incarcerated male adolescents: A randomized controlled trial of two interventions. Journal of Sex Education and Therapy, 24, 9-17. (NR)

Thoresen, C.E., Friedman, M., Gill, J.K., and Ulmer, D.K. (1982). The recurrent coronary prevention project. Some preliminary findings. Acta medica Scandinavica. Supplementum, 660,172-92. (NR)

van Bastelaar, K.M., Pouwer, F., Cuijpers, P., Twisk, J.W., and Snoek, F.J. (2008). Web-based cognitive behavioural therapy (W-CBT) for diabetes patients with comorbid depression: design of a randomised controlled trial. BMC psychiatry, 8, 9. (NR)

Cochrane Reviews (n=1):

(R=1, NR=0)

Kenworthy, Tracy., Adams, C. E., Bilby, C., Brooks-Gordon, B., and Fenton, M. (2008). Psychological interventions for those who have sexually offended or are at risk of offending. Cochrane Database of Systematic Reviews: Reviews, 4, John Wiley & Sons, Ltd Chichester, UK. (R)

Other Reviews (n=3):

(R=2, NR=1)

Bilby, C., Ferriter, M., Jones, H., Huband, N., and Smailagic, N. (2008). Psychological interventions for those who have sexually offended or are at risk of offending. Cochrane Database of Systematic Reviews: Protocols 2008 Issue 4 John Wiley & Sons, Ltd Chichester, UK. (R)

Bilby, C., Brooks-Gordon, B., and Wells, H. (2006). A systematic review of psychological interventions for sexual offenders - II: quasi-experimental and qualitative data. Journal of Forensic Psychiatry and Psychology, 17, 467-484. (R)

Walker, D. F., McGovern, S. K., Poey, E. L., and Otis, K. E. (2004). Treatment effectiveness for male adolescent sexual offenders: a meta-analysis and review. Journal of Child Sexual Abuse, 13, 281-293. (NR)

Total = R=5, NR=37

APPENDIX 2 – SEARCH STRATEGIES

Database	Search Strategy (limit to full text		
	and English language)		
Cochrane	1. drop* out* in Title, Abstract or Keywords and sex* offend* in Title, Abstract or Keywords in all products 2. treat* adhere* and sex* offend* in all products 3. attrition in Title, Abstract or Keywords and sex* offend* in Title, Abstract or Keywords in all		
	products 4. personality* in Title, Abstract or Keywords and treat* adhere* in Title, Abstract or Keywords in all products 5. personality in Title, Abstract or Keywords and sex* offend* in Title, Abstract or Keywords in all products		
	6. risk* assess* in Title, Abstract or Keywords and sex* offend* in Title, Abstract or Keywords in all products 7. key dev* variable* and sex offend* (0 results). 8. key dev* variable* and personality* (0 results.		
Embase (1980-2008)	1. Treatment Outcome/ or Patient Compliance/ or drop\$ out\$.mp. or Personality Disorder/		
	2. limit 1 to (full text and english language)		
	3. Criminal Behavior/ or Offender/ or Sexual Crime/ or Child Sexual Abuse/ or sex\$ offend\$.mp. or Rape/ or Recidivism/		
	4. limit 3 to (full text and english language)		
	5. 2 and 4		
	6. treat\$ adhere\$.mp.		
	7. limit 6 to (full text and english language) 8. 4 and 7		
	9. attrition.mp.		
	10. limit 9 to (full text and english language)		
	11. 4 and 10		
	12. personality\$.mp. or PERSONALITY DISORDER/ or PERSONALITY/		
	13. limit 12 to (full text and english language)		
	14. 7 and 13		
	4 and 13		
	15. Risk Assessment/ or risk* assess*.mp.		
	16. limit 16 to (full text and english language)		
	17. Criminal Behavior/ or Sexual Crime/ or Child Sexual Abuse/ or sex\$ offend\$.mp. or		
	Rape/ or 18. 18. Recidivism/		
	19. limit 18 to (full text and english language)		
	20. 4 and 17		

	21. dev\$ variable\$.mp.
	22. limit 21 to (full text and english language)
	23. 4 and 22
	24. 13 and 22
	25. personality disorder\$.mp. or exp Personality
	Disorders/
	26. limit 25 to (full text and english language)
	27. risk factor\$.mp. or exp Risk Factors/
	28. limit 27 to (full text and english language)
	29. 26 and 28
	30. 4 and 29
	Searches returned 12 journals relevant by title
PsychINFO (1967 ó 2008)	1-30. Search terms from Embase search history were re-executed using OVID.
	Searches returned 25 relevant journals by title
Swetswise (year: all, showing: all	1. Treatment outcome and offend
subscriptions, language: English	
categories: all)	

APPENDIX 3 - KEY DEVELOPMENTAL VARIABLES

Sexual victimisation in childhood

Sexual victimization was defined as sexual contact with another person that was either unwanted or perpetrated by an adult at least five years older than the subject. Consenting sexual contact with peers was coded separately.

Physical victimisation in childhood

Ratings of physical abuse in childhood were defined as physical contact, perpetrated by an adult on a number of occasions, which was unprovoked, or excessive in relation to any misdemeanor committed by the subject.

Emotional abuse/physical neglect in childhood

Ratings of emotional or physical neglect in childhood were defined as persistent and marked failures on behalf of the caring adult(s) to provide adequate and consistent care.

Childhood 'disturbance'

A number of variables were considered, which are associated with emotional or conduct disorder in childhood. These included ratings (before the age of 16) for persistent truanting or school refusal, significant episodes of being bullied or bullying others, suspension from school for aggression, stealing, running away from home, deliberate self harm, experiencing prolonged difficulties with peer friendships and marked feelings of misery. Subjects were defined as having experienced childhood disturbance/difficulties if they reported two or more of the above.

Trauma

One or more or two or more trauma@s are defined as having one or more or two or more of the variable@s -physical abuse, sexual abuse, emotional/physical neglect in childhood@

APPENDIX 4- MEASURES

MCMI-III (The Millon Clinical Multiaxial Inventory, Third Edition) (Millon, 1994)

The MCMI-III is a 175 item test, which is designed for use with adults aged 18 years and older. The MCMI-III supports psychologists and other qualified mental health professionals in clinical, counselling, medical, forensic and other settings in which individuals are being assessed for emotional, behavioural, or interpersonal difficulties. This instrument can be used to:

- Assess the interaction of Axis I and Axis II disorders based on the DSM-IV classification system
- Identify the deeper and pervasive personality characteristics underlying a patient's overt symptoms
- Gain an integrated understanding of the relationship between personality characteristics and clinical syndromes to facilitate treatment decisions.

The test incorporates 14 Personality Disorder Scales (11 Moderate Personality Disorder Scales and 3 Severe personality Pathology Scales), 10 Clinical Syndrome Scales (7 Moderate Syndrome Scales and 3 Severe Syndrome Scales), 4 Corrections Scales (3 Modifying Indices and 1 Validity Index).

Stable-2007 (Hanson & Harris, 2000)

The Stable-2007 is a sex offender risk assessment tool designed to measure stable dynamic risk domains to supplement static risk assessment measures. The instrument includes 13 relatively stable factors, significant social influences, relationship instability, emotional identification with children (child abusers only), hostility to women, general social rejection, lack of concern for others, impulsivity, poor problem solving skills, negative emotionality, sexual preoccupations, sex as coping, deviant sexual preferences, co-operation with supervision. The items are rated as 0 (absent), 1 (maybe present) and 2 (present). All items are then calculated to arrive at an overall Stable-2007 score. The Stable-2007 has achieved AUC of .77 (Hanson, Harris, Scott, and Helmus, 2007). The tool has been designed for use with the Static-99 and the

Acute-2007 to provide a comprehensive risk assessment. However, it can be used alone for clinical and research purposes.

Risk Matrix 2000 (Thornton et al, 2003)

The Risk Matrix 2000 (RM2000) is an evidence-based actuarial sexual violence risk assessment tool, for men over 18 with at least one conviction for a sexual offence. The tool can be used to predict the likelihood of reconviction for a sexual or violent offence in the long term (up to 15 years). It involves a two stage process, whereby in stage one three static items are scored and a risk category is determined and in stage two there are four aggravating factors, if two aggravating factors are present the risk is raised one category and if all four are present the risk is raised two categories. This tool is widely used in the Prison, Probation, and Police services in England and Wales. It has achieved AUC ranging from low 0.7¢s to low 08¢s (Thornton et al, 2003).

Static-99 (Hanson & Thornton, 2000)

The Static-99 is a brief actuarial risk assessment designed to estimate the probability of sexual and violent recidivism in adult males convicted of at least one sexual offence. The tool contains ten items associated with four broad categories: Sexual Deviance, Range of Potential Victims, Persistence, and Anti-sociality. The Static-99 is widely used in the United States and Canada and has achieved AUC of 0.71.

Psychopathy Checklist Revised (PCL-R) (Hare, 1991)

The PCL-R is a rating scale designed to measure traits of psychopathy in forensic populations. The measure contains 20 items each reflecting a different characteristic of psychopathy, which are rated on a three-point scale, from the item does not apply through to the item definitely applies A score between 0 and 0 is calculated with a cut off of 25 (U.K. samples) or 30 (U.S. samples) indicating a diagnosis of psychopathy. The measure can also be rating according to two factors, Factor one, which reflects the affective and interpersonal features, and Factor two, which reflects the social deviance features. It is designed to be completed on an interview basis combined with file information. The PCL-R has good internal consistency, inter-rate and test re-test reliabilities ranging from .85 to .95.

APPENDIX 5 - STABLE-2007 RATING SHEET

Subject Name: Place of Scoring:				
Scoring Item	Notes	Section Total		
Significant Social	110105	10111		
Influences				
Capacity for				
Relationship				
Stability				
Emotional ID with	(Only score this item for child molesters)			
Children				
Hostility toward				
women				
General Social				
Rejection				
Lack of concern for				
others				
Impulsive				
Poor Problem				
Solving Skills				
Negative				
Emotionality				
Sex Drive				
Sex Preoccupation				
Sex as Coping				
Deviant Sexual				
Preference				
	ests in Possible Remission An offender who			
	ed upon historical facts can have their			
	est score reduced by one point if the			
	The offender is involved in an age			
,	ual, satisfying sexual relationship of at least			
	while "at risk" in the community with the ral indicators of Deviant Sexual Interest for 2			
years.	ai mulcators of Deviant Sexual interest for 2			
	s relationship has been confirmed by a			
	t, collateral contact and the above condition			
· · · · · · · · · · · · · · · · · · ·	r and count a "negative 1" in this score box –			
	r's overall score by "1"			

Co-operation with

Supervision		
(Out of 24 for the	Sum for Final Total ose without a child victim, see Tab 8, page 36 for definition of a õchildö)	26

Interpretive Ranges: 0-3 = Low, 4-11 = Moderate, 12+ = High

APPENDIX 6 – PERSONALITY DISRODER CLUSTERS

Cluster A

Described as the Odd or Eccentric disorders, consists of Paranoid, Schizoid, and Schizotypal personality disorders.

Cluster B

Consists of Antisocial, Borderline, Histrionic, and Narcissistic personality disorders.

Cluster C

Consists of Obsessive-compulsive, Avoidant, and Dependant personality disorders.